Lending relationships and the real economy: evidence in the context of the euro area sovereign debt crisis

Working Papers 2017



BANCO DE PORTUGAL



# 8

Lending relationships and the real economy: evidence in the context of the euro area sovereign debt crisis

Working Papers 2017

Luciana Barbosa

June 2017 The analyses, opinions and findings of these papers represent the views of the authors, they are not necessarily those of the Banco de Portugal or the Eurosystem.

Please address correspondence to Banco de Portugal, Economics and Research Department Av. Almirante Reis 71, 1150-012 Lisboa, Portugal T +351 213 130 000 | estudos@bportugal.pt



Lisbon, 2017 • www.bportugal.pt

WORKING PAPERS | Lisbon 2017 • Banco de Portugal Av. Almirante Reis, 71 | 1150-012 Lisboa • www.bportugal.pt • Edition Economics and Research Department • ISBN 978-989-678-522-2 (online) • ISSN 2182-0422 (online)

# Lending relationships and the real economy: evidence in the context of the euro area sovereign debt crisis

# Luciana Barbosa Banco de Portugal

## June 2017

#### Abstract

The recent euro area sovereign debt crisis put the financial sector under pressure and imposed several challenges, mainly in the countries most affected by the crisis. The sovereign-bank linkage can negatively affect the economic activity, especially by bankdependent firms. This study explores the heterogeneity across banks in their funding structure, sovereign exposures, solvency, and availability of collateral, with the aim of investigating the effect of the crisis on firms' investment and employment decisions. Exploring a detailed database that covers virtually all bank loans granted to Portuguese firms, for the period 2007-2012, the results suggest an impact on investment and employment paths for firms whose lenders depend more heavily on interbank and market funding. Moreover, the results also stress the importance of assets eligible as collateral in monetary operations conducted by Central Bank. The findings suggest how a deterioration in sovereign creditworthiness can affect the real economy via the banking sector.

 $JEL:\ G21,\ G31,\ E22,\ E24,\ E44,\ E51$ 

Keywords: Sovereign debt crisis, heterogeneity firm's lenders, firm's investment and employment.

Acknowledgements: I am grateful to Diana Bonfim, Geraldo Cerqueiro, Hugo Reis, Miguel Ferreira, Paula Antão, Rui Albuquerque and Steven Ongena for their comments and discussion. I would also like to thank the seminar participants at Nova Research Group and at internal seminar at Banco de Portugal.

The analysis, opinions and findings of this paper represent the author' views, which are not necessarily those of the Banco de Portugal or the Eurosystem. All errors are author's responsibility.

E-mail: lsbarbosa@bportugal.pt

# 1. Introduction

Financial intermediation ensures the flow of capital from savers to firms (or other agents), which is crucial for economic activity and growth. The recent financial and sovereign debt crises implied severe dysfunction in the international financial markets, with repercussions on financial institutions. These events raise the discussion of how the crisis affects financial intermediaries' ability to grant credit, and emphasize the importance of understanding how shocks to credit suppliers affect the real economy.

The financial crisis in 2008 imposed huge losses for financial institutions worldwide, and led to a dry-up in the interbank markets. Later, the Greek bailout in mid-2010 marked the onset of the sovereign debt crisis in the euro area. The unprecedented and unexpected nature of this event changed the assessment and perception of sovereign credit risk by market participants. The sovereign bond yields of other euro area countries increased considerably, especially for Ireland, Portugal, and to a lesser extent, Spain and Italy, while other financial markets were also affected (*e.g.* Benzoni *et al.* (2015)).

The sovereign debt crisis and the tensions in financial markets in general were transmitted to credit institutions through several channels. A direct link was the negative impact on the net worth of sovereign debt securities held by institutions. These losses weakened the balance sheets position, which made those institutions appear to be riskier. Sovereign debt market developments also fueled the perception that Governments would have lower financial ability to support the national banking systems if needed. This too had severe implications for financial institutions. Indeed, the sovereign-bank linkage implied a marked increase in the funding cost for the institutions hosted or exposed to the stressed countries. Some institutions even lost access to the international financial markets in this period. This environment imposed several challenges to financial institutions and their activity.

This study investigates the effects of the sovereign debt crisis and financial market disruptions on corporate decisions in Portugal, namely firms' investment and employment, exploring the heterogeneity of firms' lenders. In particular, we compare investment and employment outcomes of firms that borrow from financial institutions with heterogeneous exposures to the sovereign and financial market developments.

Portugal figures as an interesting case study for several reasons. First, Portugal was at the core of the sovereign debt crisis in the euro area (Figure A.1, in the Appendix Section of this Chapter), which led to its rescue via International Financial Assistance in April 2011. Second, there were severe negative consequences on the Portuguese banking system, driven by the sovereign-bank linkage. Due to the increasing tensions and risk aversion in financial markets, Portuguese banks have faced daunting liquidity challenges since 2010 (Figure A.2, in the Appendix Section).

This fact is especially relevant as the banking system plays a critical role as a funding source to the Portuguese economy, notably to the corporate sector.<sup>1</sup> Portuguese firms, comprising mainly small and medium sized-firms (SMEs), present high leverage ratios, making them more vulnerable to changes in credit institutions' financial positions. Moreover, SMEs have less access to alternative financial instruments. Finally, it is worth mentioning that the initial shock was exogenous to the Portuguese banking system, and not driven by developments in the corporate sector.<sup>2</sup> Indeed, the increases in the Portuguese sovereign bond yields, in opposition to other countries, such as Ireland and Spain, were not driven by the Government support to the banking system or a price bubble in the real estate market. In fact, it was related essentially to the higher concerns, following the Greek bailout, related to the Portuguese macroeconomic imbalances, namely the weakness of Portuguese public finance (excessive debt levels and high deficits). These factors are important in the analysis, as banks did not anticipate the developments recorded from mid-2010 on, and the resulting loss of access to the international wholesale debt markets.

This study contributes to the literature that analyzes the impact of banksovereign linkage on a firm's decisions. To perform the analysis, we use detailed micro databases for Portuguese firms and financial institutions, which allows us to match firm-bank, to explore the intensity of these relationships, and to cover different segments of the corporate sector. We investigate if there are differences in firms' investment and employment decisions based on firmlenders relationships and the characteristics of respective lenders. Namely, we characterize firms' lenders, based on several key indicators, and identify those relationships that could be more vulnerable to the negative shock recorded in international financial markets in 2010. To the best of our knowledge, the paper most similar to ours is the recent work of Bottero *et al.* (2015). However, in this study we explore alternative channels of transmission from banks to firms that may be helpful in identifying banks more vulnerable to adverse financial developments.

According to the results obtained, the key lenders' characteristics that affect firms' decisions are related to banks' market funding positions, with a negative effect on both investment and employment. As the dependence of banks on these funding sources increases, their borrowers tend to present lower investment and employment paths. In turn, assets eligible as collateral in monetary operations seem to have a favorable impact on both firms' decisions.

<sup>1.</sup> In Portugal, and broadly in Europe, banks play an important role in financial intermediation, in contrast with other economies, such as that of the USA, where wholesale markets are also important, as discussed in Langfield and Pagano (2015).

<sup>2.</sup> As mentioned in Banco de Portugal (2011b), "Portuguese banks' liquidity difficulties resulted, to a large extent, from a contagion effect deriving from disturbances in sovereign debt markets and not directly from intrinsic problems of solvency or profitability".

Looking at banks' sovereign debt exposures and regulatory capital ratios, the results are mixed.

This Chapter is structured as follows: Section 2 presents a brief review of related literature. Section 3 presents the main facts regarding the Portuguese banking system under the International Financial Assistance Programme to Portugal. Section 4 describes the data sources, and the data set used in the analysis. Section 5 shows the empirical strategy adopted and presents some summary statistics for the variables under analysis. Section 6 presents the empirical specification and the econometric results. Section 7 explores an alternative empirical approach. Section 8 shows some robustness tests. Finally, Section 9 presents the main conclusions.

# 2. Related Literature

The value of the banking system and its impact on the real economy is not a new topic in economic and financial literature. For instance, Bernanke (1983) discussed the relevance of banks' balance sheet channel. He showed that shocks to banks' financial positions affect lending and consequently borrowers' decisions and real activity. Holmstrom and Tirole (1997) presented a model of financial intermediation in which firms and intermediaries could be capital constrained. They emphasized the role of the financial intermediaries, in addition to the wholesale market, showing that some firms have access only to external funds through those institutions or given their monitoring function. Moreover, the authors show how changes in banks' capital positions affect their credit supply, which is particularly important to less capitalized firms.

The link between the real economy and the financial sector was also emphasized in papers related to the so-called "financial accelerator". This literature argues that due to the existence of imperfections in the credit markets the general financial conditions account for the intensity and persistence of the economic business cycles (e.g. Bernanke and Gertler (1989) and Bernanke et al. (1999)). The asymmetric information on credit markets is especially important to smaller, younger, or less transparent borrowers, contributing to a greater sensitivity of this segment of firms to changes in the credit supply (e.q.Mark Gertler (1994)). King and Levine (1993), for instance, found evidence that the financial system can promote economic growth. Namely, financial developments are related to GDP growth, physical capital accumulation, and efficiency, as well as to the future evolution of these variables. The credit market imperfections and the impact of financial frictions/conditions on firms' decisions have also been a central topic in corporate finance research, both in theoretical and empirical perspectives (e.g. the seminal paper Fazzari et al. (1987), or Love (2003)).

Given its place in the financial system, bank credit has attracted intense interest in the financial literature, notably the firm-bank relationships. According to this literature, in the borrowers' perspective, there is evidence that the number of lending relationships and the length of these relationships may affect the availability of credit and contracts conditions, e.g. Petersen and Rajan (1994), Ongena and Smith (1998), Boot (2000), and Berger and Udell (2006). An important point in this literature is the acquisition of soft information through repeated interactions between borrowers and lenders (e.q. Diamond (1984)), which helps to minimize asymmetric information issues. Nevertheless, this information acquisition, and the reliance on only a few lenders, may also contribute to hold-up problems for firms (for instance, information rents, as explored by Rajan (1992)), or switching costs (as discussed in Kim et al. (2003), and recently in Chodorow-Reich (2014)). Under financial distress episodes, Hoshi et al. (1990) showed that Japanese firms with a main bank-lending relationship have been found to obtain lower costs of overcoming those events. Bae et al. (2002) explored adverse events that affected the Korean banking system during the Asian crisis in the late 1990s, and showed that adverse shocks to a firm's main lender have a negative knock-on effect not only on the value of the bank but also on the value of the firm itself.

More recently, the literature on financial-real economy linkage recorded a new wave, exploring the impact of the financial and the euro area sovereign debt crises on credit institutions and firms' decisions. While the financial crisis directly affected banks' financial health and the functioning of the interbank markets, the sovereign debt crisis may have affected financial markets and the financial institutions through several channels. Sovereign debt tensions had a direct negative impact on the market value of sovereign debt securities. Moreover, financial systems were also perceived as more vulnerable as the sovereign capacity to provide financial assistance decreased. In this context, banks' funding costs also increased. In a second round, the increases in sovereign yields may have induced changes in banks' decisions, contributing to portfolio adjustments, such as an increase in sovereign holdings for less risk averse institutions. These securities presented higher returns (which improve profitability), while they did not imply additional capital needs (zero risk weights in terms of capital requirements). This strategy may reinforce the bank-sovereign linkage. It may also imply a decrease in credit supply to other economic segments (*i.e.* a crowding out effect).

Some of the most recent research has assessed the impact of the crisis on banks' credit supply. Empirical evidence suggests a decrease in credit to firms, due to negative shocks in financial markets. Iyer *et al.* (2014) analyzed the impact of the financial crisis in 2008 on the credit supply in Portugal, exploring data from the Central Credit Register. They found that banks more dependent on the interbank market restricted credit to firms to a greater extent than did banks less exposed to that market. Bofondi *et al.* (2013) investigated a similar research question for Italy during the sovereign debt crisis. Based on the distinction between foreign and domestic banks, as the latter were affected by Italian sovereign yield increases, they found that domestic banks decreased

credit supply more than did the foreign ones. In turn, Popov and Horen (2015), and Adelino and Ferreira (2016) centered their analyses on the impact of the sovereign debt crisis on credit, evaluated through syndicated loans. Popov and Horen (2015) showed that banks with greater exposures to stressed countries recorded greater credit cuts. Exploring banks' rating downgrades as a consequence of respective sovereign ratings revisions, Adelino and Ferreira (2016) found that those banks revealed a greater impact on credit supply than did institutions that were not subject to this effect.

Another strand of the recent literature explores the potential impact of recent crises on firms' decisions, given the impact on the financial system and the relevance of bank credit as an external funding source to the corporate sector, in particular in Europe. The variables of interest are related to real decisions, such as employment and investment, as well as financial indicators, as leverage and sales growth. In general, the results suggest that there are differences in the path of firms' outcomes between firms less and more exposed to the financial and sovereign debt crises through their lenders. Based on syndicated loans, Chodorow-Reich (2014) first found that less healthy banks (exploring several metrics) reduced more credit than other banks during the financial crisis in the US.<sup>3</sup> He also found that firms that had pre-crisis relationships with weaker banks reduced more employment than did firms whose lenders were healthier. Similarly, Bentolila et al. (2015) found an impact on Spanish firms' employment policies. Firms that relied on weaker banks (in this setup, bailout banks) showed greater employment drops than firms with relationships with healthier banks. Also in the context of the financial crisis, Cingano et al. (2013) found that Italian firms whose lenders were more dependent on interbank funding reduced their investment more than firms less dependent on such banks. In the context of the European sovereign debt crisis, Bottero et al. (2015), also based on Italian data, found that banks with higher exposures to Italian sovereign debt tightened more credit supply to firms. Moreover, they found that smaller and riskier firms were not able to overcome this fact, recording a reduction in investment and employment. De Marco (2016) and Acharya et al. (2016) analyzed the impact of sovereign debt crisis on firms' decisions, exploring syndicated loans data. On average, firms whose lenders were more exposed to sovereign debt of stressed countries (in De Marco (2016)) or firms whose lead lender was from those countries (in Acharya et al. (2016)) presented a more adverse path for some firms' outputs than firms with other lending relationships.

The impact of bank credit on firms' decisions may also depend on firms' ability to substitute bank relationships or/and bank credit with other funding sources. Adjustments in firms' debt components may minimize the effects of

<sup>3.</sup> Banks' position were assessed by alternative measures related to exposures to Lehman Brothers, exposures to toxic mortgage back securities, and some balance sheet indicators.

bank credit shocks. Some papers have explored this dynamic, but the empirical results are mixed. Becker and Ivashina (2014) and Adrian et al. (2012), exploring debt market as alternative funding sources, argued that there were no real effects that could be related to banks' lending paths. In turn, Carvalho et al. (2015) found that the access to public debt markets did not offset the impact of bank distress on firms' decisions. In turn, Almeida et al. (2017) explored the direct negative spillovers of sovereign rating downgrades on firms' ratings, which has a negative effect on firms' funding costs. Their results suggest that firms that recorded a rating downgrade due to the sovereign ceiling policy, *i.e.* firms should not present higher ratings than the respective sovereign (firms' downgrade were not directly related to firms' fundamentals), showed greater impact on their decisions than did the other firms.<sup>4</sup> However, as mentioned, public debt markets are not available for all firms. Even firms that try to adjust funding within the banking system may face some important constraints. For instance, Ivashina and Scharfstein (2010) found evidence that borrowers of weaker banks could not switch to healthier banks during the financial crisis.

The present study contributes to the empirical literature that explores how the sovereign debt crisis affected financial intermediaries and corporate decisions. Looking at empirical literature, there are some papers with similarities to this analysis. Some of them analyze a similar time window, namely the euro area sovereign debt crisis, while others explore analogous databases, with special emphasis on the Central Credit Register. This database avoids the bias to larger firms that characterize some studies, such as those based on syndicated loans. Indeed, small and medium firms (SMEs) are a significant fraction of the corporate sector, and account for much of the economic activity and employment in several countries, such as Portugal. SMEs usually do not have access to the syndicated loan markets. They are typically more dependent on bank credit, and consequently more vulnerable to changes in bank credit supply. This study is also in line with papers that explore corporate decisions.

Combining all these features, to the best of our knowledge, the paper most similar to this one is the recent work of Bottero *et al.* (2015). The two papers investigate the impact of the sovereign debt crisis, exploring the Central Credit Register. This database allows a direct firm-lender match, and simultaneously an exploration of corporate heterogeneity. However, in this paper we directly explore several dimensions of a firm's lenders that may be relevant in the environment under analysis. Accordingly, in this study possible channels other than the direct exposure to sovereign debt securities are explored in more detail. For instance, we examine the structure of banks' liabilities, and especially the availability of collateral to gain access to the monetary operations conducted

<sup>4.</sup> In the Portuguese case there are few firms with rating notes and access to the wholesale funding. As a result, this direct impact is not sizeable.

by the ECB, which has not deserved much attention in the literature in this context.

# 3. The Portuguese International Economic and Financial Assistance Programme: Main facts on the banking system

The international financial crisis following the US sub-prime mortgage crisis and the collapse of Lehman Brothers had little direct impact on the Portuguese banking system as a whole. In general, banks were not exposed to the subprime market and their exposures to "toxic assets" were contained. Moreover, unlike other economies, Portugal did not record a bubble in the real estate market. Nevertheless, Portuguese institutions were affected by changes in financial market conditions, in particular by the dry-up in the interbank market during this period. Those constraints were minimized by monetary operations conducted by Central Banks and by issuing bonds with government guarantees (Figure A.3, in the Appendix Section). As a result, lending to non-financial corporations continued to grow at high rates in Portugal during this period (Figure A.4, in the Appendix Section).

However, the sovereign debt crisis marked the onset of a new period that saw several deleterious effects on the Portuguese economy and the banking system. With the Greek bailout and the increasing tension in sovereign debt markets in the euro area, there was a reassessment of sovereign credit risk by market participants. The yields of Portuguese government bonds rose dramatically. The sovereign-banking system link and the risk aversion in financial markets posed several challenges to Portuguese financial institutions. Given the increased weight of the international financial markets in the funding structure of Portuguese banks since the early 2000s (as a consequence of the financial integration in the context of the monetary union), and the exposure of Portuguese banks to sovereign debt securities, these developments required sizable adjustments in banks' funding and business strategies.

Due to the renewal of tensions in the European sovereign debt markets, which led to an escalation of the Portuguese bond yields, since end-2010, the Portuguese Government requested international assistance in April 2011. This led to the International Economic and Financial Assistance Programme (hereinafter Programme), defined for a horizon period of three years, and provided by the International Monetary Fund, the European Union, and the European Central Bank. The Programme focused on three main pillars: structural reforms and competitiveness of the Portuguese economy; fiscal consolidation; and deleverage of the financial and private sectors.

Looking at the banking system, the Programme sought to ensure an orderly and gradual deleveraging process and the reinforcement of regulatory capital positions. Simultaneously, a close assessment of the financial conditions in the economy was to be conducted, in order to ensure an equilibrium between the necessary deleveraging adjustment and the financial support to the economic activity. Three fundamental dimensions should be highlighted: i) the implementation of measures to ensure sufficient liquidity in the banking system; ii) the design of funding and capital plans for short and medium terms, to monitor the gradual deleveraging, the reduction of funding from the Eurosystem, and the path of capital needs; iii) the reinforcement of capital positions.

In order to achieve a stable funding structure, the Programme set specific targets for some key indicators. For instance, the Programme established a gradual convergence to 120 per cent of the loan-to-deposit ratio.<sup>5</sup> As far as regulatory capital was concerned, the Programme imposed higher minimum levels to the Core Tier 1 ratio, namely 9 per cent by the end 2011 and 10 per cent by the end of 2012.

The Programme included a backstop facility of 12 billion euros to the financial system (out of the 78 billion euros included in the Programme), in order to face potential capital needs, due to the new capital requirements and the adverse economic and financial environment that was foreseen during the horizon period of the Programme. Note that low capital ratios, *i.e.* close to the minimum regulatory threshold, may have a direct impact on a bank's activity.

In parallel, due to the general tensions in the sovereign debt markets in the euro area and the exposure of the European banks to sovereign assets, in 2011 the European Banking Authority (EBA) imposed the so-called "sovereign capital buffer" on the major banks in the European Union.<sup>6</sup> The "sovereign capital buffer" was computed taking into account banks' sovereign debt portfolios and the respective market value assessed in September 2011. This buffer was to be in place by the end of June 2012. These new rules imposed additional capital needs on some Portuguese banks.<sup>7</sup>

Therefore, banks had to manage their capital positions in order to meet all the new capital requirements. Against this background some banks realized significant capital increases over these years. Some of them applied to the financial system facility included in the Programme, namely BCP, Banco BPI (mostly due to the "sovereign capital buffer"), and Banif. CGD also increased its capital significantly, but in a different set up, given that CGD is a stateowned bank. Additionally, in this demanding environment banks were forced to adjust their activity strategies.

<sup>5.</sup> However, these targets were revised during the Programme, and they were replaced by guidelines aiming at a stable funding structure.

<sup>6.</sup> European banks included in the stress tests exercise conducted by the EBA.

<sup>7.</sup> In Portugal four banks were subjected to EBA's rules, namely CGD, Banco BPI, BCP, and ESFG.

# 4. Data and descriptive statistics

# 4.1. Data sources

The data set used in this study combines three different micro databases, available at Banco de Portugal, namely Central Credit Register (CRC), Bank Supervisory Data, and the Central Balance Sheet Database.

The CRC contains information on all credit granted by financial intermediaries operating in Portugal. CRC includes information on the outstanding amounts, as well as information regarding credit overdue events for each borrower, among other loan characteristics.<sup>8</sup> Institutions are required to report this information on a monthly basis to Banco de Portugal. Given the low lending threshold required for this report (50 euros), this database affords high coverage of the credit granted by the banking system to the corporate sector. It also allows identifying firm-bank lending relationships at each moment and the exposure of each institution to each firm.

The second database is the Bank Supervisory Data submitted by financial institutions to Banco de Portugal for different reference periods. This database contains financial statements for institutions operating in Portugal and prudential reports for those institutions under supervision of the Portuguese authorities. Note that some institutions, due to their typology, do not report all items.<sup>9</sup> This database allows us to obtain several financial and prudential indicators of institutions, which will be important to assess their vulnerability to financial market developments.<sup>10</sup>

For the corporate sector, we use the Simplified Corporate Information (*Informação Empresarial Simplificada* - IES), which was introduced in 2006. IES contains detailed financial data based on accounting reports, as well as other firm characteristics, such as the industry sector, age, and the average number of employees. This information allows us to characterize firms over time. It is noteworthy that IES covers virtually the entire Portuguese corporate sector. This avoids the potential sample bias that voluntary survey may introduce (the approach in place before 2006), and allows us to explore different firm segments.<sup>11</sup>

<sup>8.</sup> For further details on the CRC, see Booklet Nr.5 of Banco de Portugal (Banco de Portugal (2011a)).

<sup>9.</sup> For instance, subsidiaries of European Union institutions are not required to provide information on capital adequacy ratios to the Portuguese Bank Supervision Authority.

<sup>10.</sup> In this analysis we use data at consolidated level, taking into account that some bank decisions may be defined at the group level (such as specific portfolios).

<sup>11.</sup> Before 2006 the Central Balance Sheet Database followed a survey approach, based on economic activity criteria. After 2005, with IES, it covers virtually the entire corporate sector. For further details on the IES databases, see Supplement of Statistical Bulletin (Banco de Portugal (2008)).

# 4.2. Data set

For credit institutions we restrict the database to those classified as "Monetary Financial Institutions". Then, we collected balance sheet and profit and loss account data, allowing us to analyze the structure of assets and liabilities of institutions and their respective performance. The detailed data also allow us to determine the weight of sovereign debt securities portfolios. Based on prudential reports, we obtain the capital adequacy ratios.

In CRC, we match borrowers and all respective lenders. We define lending relationships at the banking group level, *i.e.* if a firm borrows from two institutions that belong to the same group, we define it as a single lending relationship. Then, we compute the relevance of each group to each firm, taking into account the share of credit provided by each banking group in the firm's total bank debt.

As far as IES data are concerned, we impose some criteria. First, the financial and public administration sectors were excluded. We also excluded observations with missing data for total assets, business volume, number of employees, and age. Furthermore, firms with fewer than five employees were dropped. Moreover, in order to remove outliers, we winsorize the top and bottom two per cent of the distributions of the variables under analysis. Additionally, given the purpose of this study, we collapse the corporate sample to firms that have records on the CRC.<sup>12</sup>

After the merger of the three databases and the application of the criteria, we shrunk the data set to a balanced panel data. We adopted this condition in order to analyze firms that performed their activity over the crisis period. We obtain a data set with around 219,000 firm-year observation for the period 2007-2012.

The balanced panel data implies that all firms grow more mature over the horizon period, which may have some impact on firms' outcomes. For instance, it is not expected that firms continue to present high levels of investment or employment growth over the life cycle. Other assumptions in the definition of the data set could be adopted, leading to an unbalanced panel data. However, that procedure may include other effects and events related to the financial and sovereign debt crises, for instance, the possible relationship between the financial and sovereign debt crises and firms' survival or bankruptcy episodes.

<sup>12.</sup> At this stage this criterion imposed a reduction of around 85,000 observations in the IES data set.

# 5. Empirical strategy

The empirical strategy adopted in this study proceeds as follows: we characterize credit institutions based on their financial and prudential reports. Then, we match firm-banks and compute a weighted indicator for each firm, based on the financial and prudential position of all firm's lenders. The weights applied correspond to the share of the credit granted by each lender in the firm's total bank debt. In other words, for each firm-year observation we obtain a weighted indicator based on the firm's lending relationships. The weighted scheme intends to control for the dependence of firms on each lender, *i.e.* control for the intensity of each lending relationship. Hereinafter, the weighted indicator is termed *Lenders' Indicator*. Finally, we analyze if there are significant differences in firms' outcomes, exploring firms' lenders' characteristics and the respective heterogeneity, controlling for other firms' characteristics (that may affect the outcomes).

There is no single criterion by which to classify credit institutions' vulnerability to the adverse financial market developments seen during the sovereign debt crisis. Given the nature of the negative shocks, several banks' dimensions are addressed. Due to the tensions in the international financial markets and the value of these funding sources to Portuguese institutions since the establishment of the euro area, we explore variables related to banks' liabilities structure. Therefore, looking at the funding structure, the indicators are related to banks' dependence on financial markets, customers' resources, and money market (variables computed with balance sheet data). We also assess the exposure of each institution to sovereigns by the sovereign debt securities portfolios, given the concerns related to the losses that institutions may incur due to sovereign yields increases. Moreover, we also explore the solvency position, since capital ratios are critical indicators for banks, and they may effectively constrain banks' activity. It is expected that institutions with greater capital buffers should present a greater ability to absorb unanticipated negative shocks without sizeable constraints on their activity, especially lending.

# 5.1. Variables and summary statistics

This sub-section provides descriptive statistics of variables related to firms and credit institutions included in the analysis.<sup>13</sup>

<sup>13.</sup> Note that due to missing data regarding some components, the number of observations included in the econometric analysis (presented in the next sections) may be slightly different from the figures presented below. However, this fact has not sizeable impact on major descriptive statistics.

Table 1 displays the composition of firms included in the data set. A significant fraction of the sample corresponds to micro and small firms.<sup>14,15</sup> Firm size can be a relevant indicator in the analysis, as empirical evidence suggests there are differences in the access to external funding by firms' size, usually a proxy for asymmetric information and firms' credit quality. Namely, the empirical literature suggests that smaller firms face greater constraints in obtaining external financing, which may be related to the lack of information available to external agents (less transparent firms), lower diversified activity (so, lower ability to react to unexpected negative shocks), or even lower pledgable assets. These firms are therefore the ones that are potentially more vulnerable to changes in credit supply.

	Total		By firm	ı's size:	т
	TOtal	Micro	Small	Medium	Large
2007	36,457	11,829	20,276	3,684	668
2008	36,457	11,306	20,638	3,825	688
2009	36,457	11,546	20,470	3,782	659
2010	36,457	11,412	20,531	3,832	682
2011	36,457	11,938	20,057	3,772	690
2012	36,457	13,394	18,829	3,582	652
Total	218,742	$71,\!425$	$120,\!801$	22,477	4,039

TABLE 1. Sample summary statistics

As mentioned, one of the variables of interest in this study is firm's investment. For this, we focus on yearly investment flows. Investment (INVESTMENT) is defined as the flow of investment in tangible and intangible assets of firm i in year t over the total of those assets at the end of the previous year (t-1).<sup>16</sup> The first columns of Table 2 present the path of this variable over the sample period. On average, investment presents a notable decrease in 2009, a year of economic recession in several economies, the Great Depression,

<sup>14.</sup> Firm size defined in line with the European Commission Recommendation of 6 May 2003 (2003/361/EC), Micro firms are defined as those with fewer than 10 employees and less than 2 million euro of business volume or total assets; Small firms are those with fewer than 50 employees and less than 10 million euro of business volume or total assets; Medium firms are those with fewer than 250 employees and a business volume below 50 million euros or whose total assets is lower than 43 million euros. The remaining firms are considered Large firms.

<sup>15.</sup> Recall that in the definition of the data set, we imposed some criteria. We excluded firms with fewer than five employees, which affects the micro segment.

<sup>16.</sup> The results were very similar when investment was defined based only on tangible assets.

after the collapse of Lehman Brothers at the end of 2008. From 2010 on, it continuous a downward trend.  $^{17}$ 

Table 2 also presents similar statistics for employment. IES database includes some firm characteristics in addition to financial statements, including the average number of employees. Based on this information we obtain the yearly employment change, which can be interpreted as a proxy for firm's employment decisions. Thus, the employment variable (EMPLOYMENT) corresponds to the change in the average number of employees of firm i in period t over the average number of employees at the end of the previous year (t-1). Broadly, based on mean figures, we observe a downward trend during the period under analysis.

	INV	/ESTME	NT	EMI	EMPLOYMENT					
	Mean	Median	$\operatorname{Sd}$	Mean	Median	$\operatorname{Sd}$				
2007	0.365	0.090	0.753	0.090	0.000	0.277				
2008	0.314	0.103	0.812	0.060	0.000	0.207				
2009	0.196	0.086	0.750	0.039	0.000	0.190				
2010	0.342	0.041	0.682	0.006	0.000	0.176				
2011	0.244	0.085	0.745	0.022	0.000	0.180				
2012	0.174	0.044	0.629	-0.002	0.000	0.164				

Note: INVESTMENT is defined as the flow of investment in tangible and intangible assets for each in a year over the total of those assets at the end of the previous year. EMPLOYMENT is defined as the change in the average number of employees of each firm in a year over the average number in the previous year. sd stands for standard deviation. The Mean and Median figures are based on the distribution of each variable. Note that there were some changes in the IES report in 2010, which may be underlying the evolution of investment rate between 2009 and 2010.

TABLE 2. Firm's decisions: investment and employment

Table 3 presents the summary statistics of firm-level variables that may affect firms' decisions. Those variables include profitability, sales growth, size, and the leverage of firms.

Profitability (PROFITABILITY) is defined as net earnings before provisions and depreciations over total assets. This variable captures the ability of each firm to generate funds internally, so it may be less dependent on external funding. Sales growth (SALES GROWTH) is the year-on-year change of real sales, and it is meant to control for the firm's growth opportunities.<sup>18</sup> Firm's size (SIZE) is included as the logarithm of total real assets. Size is usually related

<sup>17.</sup> From 2009 to 2010, the significant difference in the average rates should be related to a series break due to changes in IES's reports and accounting rules. Nevertheless, this was a transversal event to all firms included in the sample, so its impact should be captured by time dummies.

<sup>18.</sup> In empirical research, firm's growth opportunities are usually controlled through measures related to firm's market value. However, this approach is not possible to implement in the Portuguese case, as the share of quoted firms is very small.

to asymmetric information and credit quality. The leverage of firms is also an important dimension to control for. We therefore include the bank debt ratio (BANK DEBT), defined as bank debt over total assets.<sup>19</sup>

	Nr.	Mean	$\operatorname{Sd}$	p10	p25	p50	p75	p90
SIZE	211.741	13.78	1.45	12.11	12.82	13.63	14.58	15.63
PROFITABILITY	211,752	0.09	0.12	-0.01	0.04	0.09	0.14	0.22
BANK CREDIT	211,747 211,752	-0.03 0.66	0.26 0.31	-0.32 0.30	$-0.14 \\ 0.47$	-0.02 0.66	0.09 0.81	$\begin{array}{c} 0.24 \\ 0.93 \end{array}$

Note: sd stands for standard deviation, while p10, p25, p50, p75, and p90 stand for the percentiles 10, 25, 50, 75, and 90, respectively, of the distribution of each variable, for observations included in the econometric analysis.

TABLE 3. Sample summary statistics - Firm characteristics

Looking at financial institutions, Table 4 shows some descriptive statistics for different indicators that may identify institutions that are more vulnerable to the adverse financial market conditions, in the context of the sovereign debt crisis. Therefore, the figures correspond to the distribution of the *Lenders' Indicator*, *i.e.* firm's lenders' positions, weighted by the share of each lender in the firm's total bank debt, in the sample period.<sup>20</sup>

Concerning lenders' funding structure, the set of indicators includes the central bank funding (CENTRAL BANK), defined as central bank liabilities over total assets, interbank funding (INTERBANK), which corresponds to interbank market liabilities over total assets, and the funding in financial markets (MARKET FUNDING), defined as the wholesale debt and interbank funding over total assets. It also comprises the relevance of customers' resources, through the ratio of customers' deposits over total assets (DEPOSITS\_A), and loans over customers' deposits (LOAN-TO-DEPOSIT).

We expect to see a positive relationship between the levels of interbank and financial markets indicators and the lenders' vulnerability to market developments. Higher shares of these funding sources correspond to greater dependence on financial markets, and consequently institutions may be more exposed to the adverse developments and conditions recorded in those markets during the horizon period. For the loans-to-deposits ratio a similar rationale applies: a higher ratio means that the bank uses funding resources rather than customers' deposits (perceived as more stable funding source) to finance their

<sup>19.</sup> The definition of each variable for firms is presented in Table A.1 in the Appendix Section of this Chapter. Table A.2 presents summary statistics for some other firm characteristics included in the data set.

<sup>20.</sup> The definitions underlying each variable are presented in Table A.3 in the Appendix Section. Table A.4 presents the correlation matrix between firms' decisions and lenders' indicators.

lending activity.<sup>21</sup> In line with this perception, a negative relationship is also expected between deposit-to-assets and bank's vulnerability to financial market events. For the central bank indicator, during the sample period there is no clear "a priori" expectation. On one hand, the relationship may be positive, given that the ECB was crucial as a lender of last resort. On the other, since it may also identify banks' liquidity needs, the relationship may have the opposite sign. The central bank was an important funding source in the period, due to the constraints in access to alternative finance sources faced by institutions.

Concerning the sovereign exposures (the indicator most used in the recent empirical literature), we assess the total sovereign debt securities (SOVEREIGN), as the ratio of sovereign debt securities portfolio over total assets, and the Portuguese sovereign debt securities (PT SOVEREIGN), which corresponds to the Portuguese sovereign debt securities over total assets. Given the tensions in the euro area sovereign debt markets and the extreme increases in sovereign bond yields, institutions with greater sovereign exposures may be assessed as more vulnerable (which is in line with the EBA's decision about the "Sovereign capital buffer" in 2011). For these institutions, the bank-sovereign linkage is expected to be more important.

Finally, for the solvency position, the analysis takes into account the Total capital ratio (CAPITAL RATIO), *i.e.* the total regulatory capital over risk weighted assets. It also includes the Tier 1 capital ratio (TIER 1 RATIO), defined as the Tier 1 capital over risk-weighted assets, which became more relevant after the onset of the crisis. Based on capital ratios, institutions can be seen as weaker, *i.e.* more vulnerable, if they present ratios close to the legal threshold. This means that those institutions have lower capital to absorb unexpected negative shocks. Therefore, they have lower ability to react to those shocks without restrictions on their activity (and) or increases in their regulatory capital levels.

# 6. Empirical results

# 6.1. Empirical specification

The empirical strategy explores firms' decisions conditioned on their lenders' vulnerability to the adverse developments recorded in the financial markets. Therefore, in this Section, we run the following reduced-form specification:

$$y_{i,t} = c + \alpha X_{i,t-1} + \delta z_i + \varphi w_t + \beta_1 L I_{i,t-1} + \beta_2 L I_{i,t-1} \times Crisis + \mu_{i,t}$$
(1)

<sup>21.</sup> Indeed, in Portugal even during the crisis, customers' deposits presented a positive path, which reflected customers' confidence in the Portuguese financial system. At aggregate level customers' deposits in the Portuguese banking system increased by around 15 per cent from 2008 to 2012, mainly since 2010.

	Nr.	Mean	$\operatorname{Sd}$	p10	p25	p50	p75	p90
Funding structure:								
CENTRAL BANK	211,752	0.05	0.04	0.00	0.02	0.04	0.08	0.12
INTERBANK MARKET	211,752	0.11	0.12	0.03	0.05	0.08	0.12	0.24
MARKET FUNDING	211,752	0.27	0.13	0.13	0.20	0.27	0.35	0.44
DEPOSITS A	211,752	0.45	0.13	0.30	0.38	0.45	0.51	0.60
LOAN-TO-DEPOSIT	$211,\!752$	2.31	11.20	0.86	1.05	1.29	1.56	1.85
Debt securities portfolio:								
PT SOVEREIGN	211,752	0.03	0.03	0.00	0.01	0.03	0.05	0.06
SOVEREIGN	$211,\!752$	0.04	0.03	0.01	0.02	0.04	0.05	0.07
a .								
Solvency:								
TIER 1 CAPITAL	211,752	0.07	0.05	0.05	0.06	0.08	0.09	0.11
TOTAL CAPITAL	$211,\!752$	0.09	0.05	0.06	0.09	0.10	0.11	0.12

Note: sd stands for standard deviation, while p10, p25, p50, p75, and p90 stand for the percentiles 10, 25, 50, 75, and 90, respectively, of the distribution of each indicator. The *Lenders' Indicator* corresponds to a weighted indicator at firm level, based on share of each lender on firm's total bank debt.

TABLE 4. Descriptive statistics - Lenders' Indicators

where the left-hand-variable,  $y_{i,t}$ , corresponds to the decision of firm *i* in period t, namely investment or employment decisions.  $X_{i,t-1}$  is a vector of firm-specific variables that may affect a firm's decisions, measured at t - 1.<sup>22</sup>  $z_i$  corresponds to the firm's time-invariant components. The firm fixed-effects control for unobserved firm characteristics that are unchanged over time.  $w_t$  represents year dummies, which control for changes in the general macroeconomic and financial environment that affect all firms simultaneously. LI is the *Lenders' Indicator*, the variable that characterizes the position of all firms' lenders.  $LI_{i,t-1}$  reflects firm *i*'s lenders' position, based on the criterion under analysis, at t - 1. The specification also includes an interaction term between this variable and the time dummy variable Crisis, that takes the value one for the period after the onset of the euro area sovereign debt crisis, *i.e.* after 2009  $(LI_{i,t-1} \times Crisis)$ . Finally,  $\mu_{i,t}$  corresponds to the error term.

Based on equation 1, we are interested in the sign of the  $\beta_2$  coefficient, as it allows us to know if the sensitivity of firms' decisions to lenders' characteristics changed after 2009, when banks suffered the negative shock related to the outbreak of the euro area sovereign debt crisis, and the spillovers to Portuguese agents and economy.

As the specification includes fixed-effects at firm level, the identification comes from the comparison within firms' changes in employment and investment, for firms that borrow from lenders with different exposures to

<sup>22.</sup> The inclusion of control variables with a lag period avoids the contemporaneous effect between firm's characteristics and its decisions.

the crisis. Moreover, due to the specificities of the data set, the econometric procedure includes robust standard errors.

# 6.2. Econometric results

# i) Investment decisions

As mentioned, the purpose of this study is to explore if firms that have relationships with lenders more vulnerable to the adverse environment, *i.e.* subject to higher challenges during the sovereign debt crisis and the adverse financial market developments, present significant differences in their decisions. In this section, we focus on investment outcomes.

Investment is an important component for firm's prospects, and, at the aggregate level, it is closely related to economic growth. In fact, as shown in Amador and Coimbra (2007) and Almeida and Félix (2006), capital stock developments have made an important contribution to Portuguese economic growth in the past few decades. More recently, the low performance of the Portuguese economy has been linked to a strong fall in investment.<sup>23</sup> In general, investment is fundamental in determining the future productive capacity and economic growth in the long-run (*e.g.* King and Levine (1993)).

As mentioned above, we explore alternative indicators to characterize firm's lenders' position regarding funding structure, sovereign credit risk exposure, and solvency. We also control for the importance of each lender in the firm's total bank debt, which allows us to control for the dependence of the firm on each lender. The higher is the dependence of firms on lenders identified as potentially more vulnerable to the adverse market conditions, we expect that those firms face higher constraints, *ceteris paribus*. This is in line with hold-up issues and switching costs related to bank lending relationships, *i.e.* it may not be easy for firms to change lenders. Indeed, in the context of the financial crisis, Ivashina and Scharfstein (2010) found evidence that borrowers from weaker banks could not switch to healthier banks. It is also linked to the empirical evidence toward adjustments in credit supply during crisis periods, as shown by Chodorow-Reich (2014), Iyer *et al.* (2014), and Bofondi *et al.* (2013).

Table 5 presents the results of equation 1 for investment. Each column of the table corresponds to one of the alternative indicators that underlies the characterization of firm's lenders' vulnerability. For instance, the first five columns of the table present the results exploring the five funding structure criteria.

According to the results obtained, the coefficient of interbank funding is negative and statistically significant. The interaction term with the crisis dummy also has a negative and statistically significant coefficient. These results

<sup>23.</sup> Note that the concept of investment at aggregate level may not match with the investment measures computed at the micro level.

suggest that the firms whose lenders had a higher dependence of this funding source presented a poorer path for investment, and this effect was intensified after 2009. Looking at market funding, the interaction term also shows a statistically significant and negative coefficient. This result is in line with the constraints in the access to the wholesale debt markets that banks faced in that period, and the tensions in the interbank market. The deposits-to-asset ratio has a statistically significant coefficient for the interaction term, but with a negative sign. This result contradicts the *a priori* expectations, given that a higher ratio corresponds to lower dependence on alternative funding sources (rather than deposits) to finance lending operations. The coefficients of the other funding indicators are not statistically significant in this approach.

As far as lenders' sovereign debt securities are concerned, in columns 6 and 7, the empirical evidence does not suggest impacts on firm's investment. The coefficients of total sovereign debt securities portfolio, and the Portuguese sovereign debt securities holdings, are not statistically significant.

Columns 8 and 9 present the results based on Tier 1 and Total capital ratios, respectively. According to the estimates, there is no statistically significant impact on firm's investment influenced by lenders' capital ratios.

Summing up. The empirical evidence suggests that the main dimension that affect a firm's investment decision is related to the funding structure of their lenders. Broadly, according to the results, firms whose lenders depend more on funding obtained in the financial and interbank markets present lower investment after 2009, in comparison with the previous years. We find that, on average, a one standard-deviation increase in lenders' interbank funding leads to around 0.7 percentage points additional decline in firm's investment after the outbreak of the crisis.

The importance of lenders' funding structure is in line with Cingano *et al.* (2013), who also explored lenders' funding sources, even though during the financial crisis. However, these results are unlike those in De Marco (2016), who explored the sovereign debt crisis based on banks' exposures to sovereigns. However, the analysis was based on syndicated loans, which are usually biased to larger firms, and consequently may have influence in the analysis.

Firm control variables. Finally, all firms' control variables included in the specifications (the  $X_{i,t-1}$  vector) are statistically significant and present the expected sign. This confirms that firm characteristics are also important factors underlying firms' investment. In particular, SIZE shows a negative coefficient, suggesting that larger firms tend to present lower investment, which should be in line with the firm's life cycle (*i.e.* typically investment is stronger in initial phases of firms). SALES GROWTH, a variable that seeks to capture the potential growth, shows a positive coefficient. This suggests that corporate investment is sensitive to the demand for firms' products and opportunities to expand. Additionally, firms with higher indebtedness level (BANK DEBT)

tend to present lower investment in the following year. A possible reason for this result may be the fact that firms with higher debt ratios tend to face higher financial constraints. In turn, PROFITABILITY presents a positive and statistically significant coefficient. This suggests that firms use part of internal funds for investment (which is in line with the "*pecking order theory*" of Myers (1984), arguing an optimal hierarchy for funding).

# *ii)* Employment decisions

Funding is important for investment decisions, but also for other dimension of firms' activity. Therefore, we perform an analysis similar to that conducted above, but exploring firm's employment decisions. Indeed, employment is also an important firm decision.

Employment can be seen as a complementary input to production, due to the usual complementarity of physical capital and human resources in the production function. However, if firms face financial difficulties, they may also adjust this factor *per se* to levels more consistent with firms' current financial position. As described in the literature section, some papers reported evidence that firms adjust employment after a shock that affected their lenders negatively, for instance, the findings of Chodorow-Reich (2014) in the US, Bentolila *et al.* (2015) for Spanish firms, and Bottero *et al.* (2015) for smaller firms in Italy.

IES includes the average number of employees for each firm. Based on this information, we explore the potential impact of lenders' exposure to the crisis on firms' employment decisions. The econometric results for this analysis are presented in Table 6. The structure of this table follows the previous one, so each column corresponds to the alternative indicator used to characterize institutions potentially more vulnerable to the adverse financial market developments, during the sovereign debt crisis.

For lenders' funding structure, in general, firms whose lenders are more dependent on the financial market tend to present a higher adjustment in the number of employees. In particular, the coefficients of interbank and market funding indicators are negative and statistically significant after 2009. The analysis based on deposits-to-assets ratio presents a positive coefficient after the onset of the sovereign debt crisis. This result suggests that firms whose lenders finance their activity more intensively by customers' resources present a more favorable employment path, reinforcing the previous findings regarding market funding. The estimates for the remaining indicators related to lenders' liabilities are not statistically significant.

When we characterize institutions based on sovereign debt securities portfolios, both criteria (total sovereign and Portuguese sovereign securities) present positive coefficients in the period after the outbreak of the euro area sovereign debt crisis. If we expect that higher exposures imply higher risk for institutions, and consequently (negative) "differentiation" by investors in financial markets, these results are somewhat puzzling. However, this may be related to the general position of these institutions. Note that before the unexpected Greek bailout, sovereign debt securities were perceived as safe assets, and in that period the returns were high.

Based on the regulatory capital position, the Tier I and Total capital ratios are not statistically significant.

In terms of magnitudes, the estimates obtained indicate that, on average, firm's employment decreases by 0.3 percentage points based on a one standard-deviation increase in lender's market funding, after the onset of the sovereign debt crisis. In turn, a one standard deviation increase in the lenders' sovereign portfolio leads to an increase by around 0.6 percentage points in firm's employment changes.

All in all, looking at firms' employment decisions, the results are broadly in line with those observed for investment as far as lenders' funding position is concerned. However, in this analysis the results suggest that some other lenders' characteristics may also play a role on firms' employment, namely the sovereign debt securities portfolios.

		Fur	ding structu	re		Sovereig	gn exposure	Solve	ency
	1	2	3	4	5	6	7	8	9
LI:	Central bank	Interbank	Market	Dep_A	LTD	Sovereign	PT Sovereign	Tier 1 Capital	Total capital
$LI_{t-1}$	$0.1845 \ (1.07)$	$-0.0659^{**}$ (-2.02)	$^{-0.0582**}_{(-2.01)}$	$0.0403 \\ (1.47)$	-0.0010 (-1.57)	$0.4398 \\ (1.20)$	-0.0297 $(-0.11)$	-0.0911 (-1.04)	-0.0840 $(-0.97)$
$LI_{t-1}^*$ CRISIS	$-0.1272 \\ (-0.73)$	$^{-0.0627**}_{(-1.96)}$	$^{-0.0689**}_{(-2.31)}$	$^{-0.0591**}_{(-2.15)}$	-0.0004 $(-0.62)$	-0.4451 $(-1.33)$	-0.0333 $(-0.13)$	$\begin{array}{c} 0.0053 \\ (0.06) \end{array}$	$-0.0235 \\ (-0.29)$
PROFITABILITYt-1	$0.0793^{***} \\ (3.22)$	$0.0807^{***}$ (3.27)	$0.0802^{***}$ (3.25)	$0.0792^{***} \\ (3.21)$	$0.0802^{***}$ (3.25)	$0.0794^{***}$ (3.22)	$0.0791^{***} \\ (3.21)$	$0.0794^{***} \\ (3.22)$	$0.0795^{***} \\ (3.22)$
SALES GROWTHt-1	$0.1051^{***}$ (14.54)	$0.1046^{***} (14.48)$	$0.1047^{***} (14.50)$	$0.1050^{***} (14.53)$	$0.1049^{***}$ (14.52)	$0.1050^{***} (14.53)$	$0.1050^{***} \\ (14.54)$	$0.1050^{***} (14.54)$	$0.1050^{***} (14.54)$
SIZEt-1	$^{-0.5163***}_{(-49.70)}$	$^{-0.5154^{***}}_{(-49.59)}$	$^{-0.5161***}_{(-49.69)}$	$^{-0.5165***}_{(-49.64)}$	$^{-0.5162***}_{(-49.69)}$	$^{-0.5164***}_{(-49.70)}$	$-0.5165^{***} (-49.69)$	$^{-0.5165***}_{(-49.72)}$	$^{-0.5165***}_{(-49.73)}$
BANK DEBTt-1	$^{-0.3935***}_{(-22.74)}$	-0.3930*** (-22.72)	$-0.3945^{***}$ (-22.80)	$-0.3962^{***} (-22.81)$	$-0.3943^{***}$ (-22.80)	$-0.3947^{***}$ (-22.74)	$-0.3946^{***} (-22.73)$	$-0.3941^{***} (-22.79)$	$-0.3943^{***} \\ (-22.80)$
Constant	yes	yes	yes	yes	yes	yes	yes	yes	yes
rear dummies	yes	yes	yes	yes	yes	yes	yes	yes	yes
r n m-nxeu enects Nr	yes 211-752	yes 211 752	yes 211 752	yes 211 752	yes 211 752	yes 211 752	yes 211 752	yes 211 752	yes 211-752
$\mathbb{R}^2$	0.0520	0.0521	0.0521	0.0520	0.0522	0.0520	0.0520	0.0520	0.0520

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* more vulnerable institutions). The dependent variable is INVESTMENT, defined as the flow of investment in tangible and intangible assets for each firm in a year over the total of those assets at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,t-1}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt.  $LI \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010 (2010 to 2012).

TABLE 5. Firms' investment decisions

		Fui	nding structu	re		$\operatorname{Sovereis}$	gn exposure	Solve	ency
	1	2	3	4	5	6	7	8	9
LI:	Central bank	Interbank	Market	Dep_A	LTD	Sovereign	PT Sovereign	Tier 1 Capital	Total capital
$LI_{t-1}$	-0.0474 $(-1.08)$	$0.0059 \\ (0.76)$	$\begin{array}{c} 0.0030 \\ (0.41) \end{array}$	$^{-0.0123*}_{(-1.80)}$	$0.0003^{st}$ (1.85)	$-0.1408 \\ (-1.39)$	-0.0994 $(-1.42)$	$\begin{array}{c} 0.0313 \\ (1.39) \end{array}$	$0.0345 \ (1.57)$
$LI_{t-1}$ *CRISIS	$\begin{array}{c} 0.0445 \\ (1.00) \end{array}$	$^{-0.0184^{stst}}_{(-2.35)}$	$-0.0252^{***} (-3.40)$	$0.0176^{**}$ (2.52)	-0.0001 $(-0.88)$	$0.1949^{**}$ (2.10)	$0.1435^{**} \\ (2.22)$	$-0.0160 \\ (-0.75)$	-0.0258 $(-1.25)$
PROFITABILITYt-1	$0.0809^{***} \\ (12.94)$	$0.0811^{***}$ (12.96)	$0.0811^{***} (12.96)$	$0.0810^{***} (12.94)$	$0.0808^{***} \\ (12.93)$	$0.0810^{***} (12.94)$	$egin{array}{c} 0.0810^{***}\ (12.94) \end{array}$	$0.0809^{***} \\ (12.94)$	$0.0809^{***} \\ (12.94)$
SALES GROWTHt-1	$\begin{array}{c} 0.0367^{***} \ (17.35) \end{array}$	$\begin{array}{c} 0.0367^{***} \ (17.34) \end{array}$	$\begin{array}{c} 0.0367^{***} \ (17.35) \end{array}$	$0.0368^{***} \\ (17.37)$	$0.0368^{***} \\ (17.37)$	$0.0368^{***} \\ (17.36)$	$egin{array}{c} 0.0368^{***}\ (17.36) \end{array}$	$egin{array}{c} 0.0367^{***}\ (17.36) \end{array}$	$\begin{array}{c} 0.0367^{***} \ (17.36) \end{array}$
SIZEt-1	$-0.0418^{***}$ (-19.48)	$^{-0.0417^{***}}_{(-19.43)}$	$^{-0.0417***}_{(-19.47)}$	$^{-0.0417***}_{(-19.44)}$	$^{-0.0418^{***}}_{(-19.49)}$	$^{-0.0417***}_{(-19.45)}$	$-0.0417^{***} (-19.44)$	$^{-0.0417***}_{(-19.47)}$	$^{-0.0418***}_{(-19.48)}$
BANK DEBTt-1	$-0.0389^{***}$ (-8.88)	-0.0386*** (-8.83)	$-0.0387^{***}$ (-8.85)	-0.0382*** (-8.70)	$-0.0387^{***}$ (-8.85)	$-0.0382^{***}$ (-8.71)	$-0.0383^{***}$ (-8.73)	-0.0388*** (-8.87)	$-0.0388^{***}$ (-8.86)
Constant Year dummies Firm-fixed effects Nr.	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288	yes yes 212,288

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* more vulnerable institutions). The dependent variable is EMPLOYMENT, defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,t-1}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt.  $LI \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010 (2010 to 2012).

TABLE 6. Firms' employment decisions

# 6.3. Firms' decisions sensitivity over time

In this sub-section we replace the iteration term between the *Lenders' Indicator* and the crisis dummy in equation 1, by interaction terms between  $LI_{i,t-1}$  and the year dummies variables,  $w_t$ . This specification allows us to observe how the impact of lenders' characteristics changes year by year on firms' decisions. Thus, we run the following specification:

$$y_{i,t} = c + \alpha X_{i,t-1} + \delta z_i + \varphi w_t + \beta_1 L I_{i,t-1} + \beta_{2,t} L I_{i,t-1} \times w_t + \mu_{i,t}$$
(2)

Again,  $y_{i,t}$  corresponds to investment or employment decisions of firm *i* in period *t*.  $X_{i,t-1}$  is the vector of firm-specific variables at t-1, while  $z_i$  corresponds to firm time-invariant characteristics.  $w_t$  corresponds to year dummy variables.  $LI_{i,t-1}$  is the indicator that characterizes lenders of firm *i* at period t-1.  $LI_{i,t-1} \times w_t$  corresponds to the interaction term between lenders' characteristics and the year dummies.  $\mu_{i,t}$  represents the error term.

In this specification we are interested in the sign and path of  $\beta_{2,t}$  coefficients. These coefficients, in line with the previous specification, allow us to identify the impact of firms' lenders' position on firms' decisions. Additionally, we can check to see if this effect changed over the sample period. The results for this specification are presented in Tables A.5 and A.6 in the Appendix Section of this Chapter, for investment and employment decisions, respectively.

# i) Investment decisions

For investment, we confirm the relevance of firm's lenders' funding structure indicators, mainly interbank and market funding. Based on the dependence on interbank funding, the coefficients of some interaction terms are negative and statistically significant. The negative coefficient in 2009 should reflect the tensions in financial markets following the Lehman Brothers' bankruptcy, which implied a collapse of the interbank market worldwide. The coefficients are also negative and statistically significant in 2011 and 2012, showing higher magnitudes. This suggests an intensification of this lenders' characteristic during the sovereign crisis. In turn, looking at market funding, the interaction terms are statistically significant, with negative coefficients, only in 2011 and 2012, when tensions spread to several financial markets. These results are in line with the constraints that banks faced in that period after the onset of the sovereign debt crisis. The sovereign debt securities portfolios and the prudential capital positions remain statistically non-significant in this approach.

## *ii)* Employment decisions

For employment, the results also confirm the relevance of lenders' liabilities structure. Based on the interbank indicator, all the coefficients are statistically significant. This suggests that this dependence could have been stricter to banks as the crisis lasts. A similar effect is observed for the market funding indicator. In this analysis, deposits-to-assets is also statistically significant, with positive coefficients, in the last year under analysis. This may be related to significant changes recorded in this period, in particular active policies adopted by banks to capture customers' deposits, in line the guidelines defined in the Programme to the loan-to-deposits ratio. Indeed, in the beginning of the Portuguese Programme, there was a sizable decrease of this ratio through deposits effects, rather than a significant cut in lending activity. The sovereign exposures present statistically and significant coefficients in 2012, with a positive sign. The capital ratios' coefficients are positive and statistically significant in 2009 and 2012.

# 7. Alternative Approach

In this Section we intend to answer the same research question: are firm's decisions conditioned on their lenders' vulnerabilities to the sovereign debt crisis? However, we adopt an alternative empirical approach to estimate the impact of the negative shock due to the unexpected bailout of a euro area country, and its the spillovers to other countries and financial markets. In particular, in this section we explore the firm-lenders relationships at the moment precisely before the onset of the sovereign debt crisis, namely at the end of 2009. In turn, for firms' outcomes, we focus the analysis on the investment and employment average figures, before and after the onset of the crisis.

# 7.1. Alternative empirical specification

The empirical procedure of this alternative approach is similar to that followed in previous sections. Therefore, taking into account firm-bank relationships and lenders' position at the end of 2009, we compute the weighted indicator for each firm. Then, based on firms' lenders' heterogeneity, we compare the path of average investment and employment changes in the period before the outbreak of the crisis and the period after it (pre-crisis *versus* crisis). For the period before the sovereign debt shock we considered the average investment and employment based on 2008 and 2009 figures, while for the period after the shock the average variables are based on 2011 and 2012 figures.

In this section we run the following specification:

$$y_{i,t} = c + \alpha X_{i,t-1} + \delta z_i + \varphi Crisis + \beta LI_{i,09} \times Crisis + \mu_{i,t}$$
(3)

where  $y_{i,t}$  corresponds to the average investment or average employment of firm i in the period t: 2008/2009 and 2011/2012.  $X_{i,t-1}$  is a vector of firm-specific variables that may affect the firm's decisions evaluated at t-1 (namely 2007 and 2010).  $z_i$  corresponds to the firm's time-invariant components, while *Crisis* is a time dummy variable that takes the value one for the period after 2009. In this specification  $LI_{i,09}$  is the *Lenders' Indicator* that characterizes all lenders' positions of firm i at the end of 2009. Finally,  $\mu_{i,t}$  corresponds to the error term.

In this framework the coefficient of interest is  $\beta$ , which corresponds to the interaction term between the *Lenders' Indicator* and the time dummy *Crisis*. The sign and significance of the  $\beta$  coefficient allows us to check if there were differences in the firm's decisions depending on its lenders' vulnerabilities to the negative shock recorded in mid-2010. So, we test if firms whose pre-crisis lenders were more vulnerable presented higher or lower changes from the period before to the period after the onset of the sovereign debt crisis.

Given that the key coefficient reflects the interaction between the time dummy and the *Lenders' Indicator*, which is a continuous variable, we can find some similarities between this specification and those that exploit treatment intensity variables. Exploring firms variation, the set of firms whose lenders present better indicators (*i.e.* lower vulnerability to the crisis) can be defined, by analogy, as the control group. In some way, this approach is in line to the analysis conducted by Acemoglu and Lyle (2004) about the impact of World War II on the women's labor market in the US. The authors explored the "mobilization rate" of men to the war as the key differential variable, given that this rate was not uniform across states (*i.e.* the share of men recruited in each state). So, they explored a continuous treatment variable rather than the usual binary variable approach.<sup>24</sup>

In this framework, it is worth highlighting some facts. As mentioned, the key variable under analysis is defined based on lenders' positions at the end of 2009  $(LI_{i,09})$ . This position is defined after some firms' decisions had taken place. However, this fact should not invalidate the results as long as the events were unexpected at the time. In other words, before 2010 firms could not anticipate the onset of the sovereign debt crisis and the tensions in the international financial markets, or the consequences of these developments to their lenders. The same applies to financial institutions, *i.e.* institutions' decisions and financial positions at the end of 2009 did not reflect the coming events. Both arguments seem reasonable in the setup under analysis. Actually, the Greek bailout, *i.e.* the bailout of one country in the euro area, was not expected by agents and it changed considerably the assessment of sovereign credit risk by investors and the dynamics in international financial markets. Moreover, this approach allows us to exclude from the analysis potential effects related to lenders' decisions driven by the changes in the economic and financial environment, following the Greek bailout.

# Lenders' characteristics

In this approach, we continue exploring the indicators used in the previous section to characterize banking institutions. However, as we characterize firm's lenders only at the end of 2009, we include some additional indicators in the

<sup>24.</sup> For additional details and discussion about treatment intensity variables, see the Chapter "Parallel worlds: fixed effects, differences-in-differences, and panel data" in Angrist, Joshua D and Pischke, Jörn-Steffen (2008).

analysis. These indicators allow us to explore other dimensions that may help to characterize banks' vulnerability to the crisis or their potential ability to overcome the adverse market conditions.

Based on the prudential liquidity reports, we obtain information about the liquidity position of each institution, such as the liquidity gap. This indicator takes into account the maturity (mis)match between assets and liabilities for different time windows. In this analysis we consider the liquidity gap for 6 to 12 months. This indicator therefore identifies the funding needs for the second half of 2010, the semester following the Greek rescue.

As financial markets dried up for Portuguese institutions during 2010, funding from the central bank, the lender of last resort, was crucial for some institutions to overcome liquidity shocks during this period. As presented in Figure A.3, in the Appendix Section, a sizable increase in credit from ECB operations occurred in mid-2010. However, gaining access to these operations requires pledging collateral. Therefore, we include in the analysis a variable related to the availability of assets eligible for monetary operations, *i.e.* assets that could be pledged as collateral in monetary operations conducted by the ECB, if needed.<sup>25</sup>

A negative liquidity gap, LIQUIDITY GAP, means that liabilities with maturity between 6 and 12 months are higher than the assets with similar maturity. The indicator ELIGIBLE ASSETS was defined as the share of assets that can be pledged as collateral in monetary operations over total assets. Based on these indicators, we expect that institutions showing higher indicators should be in a better position to face the negative shock recorded in 2010.

Table 7 presents some descriptive statistics for *Lenders' Indicators* (in line with the analysis of Table 4), assessed at the end of 2009 and including the additional variables.

<sup>25.</sup> This key indicator is available only since 2009, with the introduction of new items in the prudential liquidity report.

	Nr.	Mean	$\operatorname{Sd}$	p10	p25	p50	p75	p90
Funding structure:								
CENTRAL BANK	36,408	0.04	0.01	0.02	0.03	0.04	0.04	0.05
INTERBANK MARKET	36,408	0.11	0.10	0.04	0.06	0.08	0.12	0.21
MARKET FUNDING	36,408	0.31	0.12	0.18	0.26	0.30	0.38	0.46
DEPOSITS A	36,408	0.43	0.13	0.29	0.34	0.42	0.49	0.54
LOAN-TO-DEPOSIT	36,408	1.42	2.47	0.91	1.16	1.33	1.50	1.79
LIQUIDITY GAP	36,408	-10.00	11.27	-19.60	-12.43	-9.66	-6.34	-1.99
Debt securities portfolio:								
PT SOVEREIGN	36,408	0.02	0.02	0.00	0.01	0.02	0.03	0.04
SOVEREIGN	36,408	0.03	0.02	0.01	0.02	0.03	0.04	0.05
Collateral:								
ELIGIBLE ASSETS	36.408	0.10	0.04	0.06	0.07	0.10	0.11	0.17
	,							
Solvency:								
TIER 1 CAPITAL	36.408	0.07	0.05	0.04	0.07	0.08	0.09	0.10
TOTAL CAPITAL	36,408	0.09	0.05	0.06	0.09	0.11	0.11	0.12
	55,100	5.00	0.00	5.00	0.00	0.111	0.111	0.12

Note: sd stands for standard deviation, while p10, p25, p50, p75, and p90 stand for the percentiles 10, 25, 50, 75, and 90, respectively, of the distribution of each indicator. The *Lenders' Indicator* corresponds to a weighted indicator at firm level, based on share of each lender on firm's total bank debt.

TABLE 7. Descriptive statistics - Lenders' Indicators at end-2009

# 7.2. Econometric results

Tables 8 and 9 present the results of this alternative approach for firms' investment and employment decisions, respectively. The new indicators are presented in the last two columns of each table.

As mentioned, the  $\beta$  coefficient in equation 3 is the key variable in this approach. Due to the characteristics of the data set, the econometrics procedure includes robust standard errors.

#### i) Investment decisions

Looking at funding structure, only the specifications with interbank and market funding indicators present statistically significant coefficients, with negative signs. These results suggest that firms whose lenders were more dependent on these funding sources at the end of 2009 presented lower (average) investment after 2010. The sovereign exposures and solvency indicators remained statistically non-significant in this set up.

Looking at the new indicators, the coefficient of the variable related to liquidity gap, presented in column 10, is not statistically significant. The availability of assets eligible as collateral, column 11, presents a positive and statistically significant coefficient for the interaction term with the crisis dummy. This suggests higher average investment for firms whose lenders showed greater ability, assessed at the end of 2009, to gain access to the ECB operations. According to the results obtained, a one standard deviation increase in lenders' market funding corresponds to a reduction of 0.7 percentage points in average investment between the period before and after the onset of the sovereign crisis. A change of similar magnitude in lenders' eligible assets indicator has approximately the inverse impact.

# ii) Employment decisions

Concerning employment decisions, and looking at lenders' liabilities structure, interbank lending, market funding, and the deposits-to-assets ratio are statistically significant. The last presents a positive coefficient while the other two indicators show negative signs. Therefore, average employment changes show less favorable path for firms whose lenders were more dependent on financial markets (or with lower share of customers' deposits) at the end of 2009.

Under this framework the indicators related to banks' exposure to sovereign debt securities continue presenting positive coefficients, *i.e.* firms whose lenders have higher sovereign debt securities holdings at the end of 2009 tend to reveal more favorable employment changes over time. Looking at capital adequacy ratios, the results do not suggest an impact on the average employment changes.

In line with the results observed for investment, the liquidity gap is not statistically significant. The variable eligible assets presents a positive coefficient. This suggests that banks with greater capacity to borrow from the Central Bank, measured by the assets available to pledge as collateral in monetary operations, are in a more favorable position to overcome adverse financial market developments.

The results indicate that firm's average employment change decreases by around 0.2 percentage points for a one standard-deviation increase in lenders' market funding. In turn, a one standard-deviation increase in lenders' sovereign securities portfolio or in eligible assets indicator implies an increase by around 0.24 percentage points in firm's average employment change.

	Funding structure					Sovereign exposure Solvency				Funding	Collateral
	1	2	3	4	5	6	7	8	9	10	11
LI <sub>09</sub> :	Central bank	Interbank	Market	Dep_A	LTD	PT Sovereign	Sovereign	Tier 1 Capital	Total capital	Liq. Gap	Elegible A
LI <sub>09</sub> *CRISIS	$0.2729 \\ (1.08)$	-0.0811** (-2.35)	$-0.0604^{**}$ (-2.08)	-0.0124 (-0.47)	-0.0016 (-0.77)	$\begin{array}{c} 0.1716 \\ (0.84) \end{array}$	$0.3287 \\ (1.57)$	-0.0493 (-0.66)	-0.0613 (-0.88)	-0.0002 $(-0.59)$	$\begin{array}{c} 0.1570^{*} \\ (1.92) \end{array}$
PROFITA BILITYt-1	-0.0116 $(-0.35)$	-0.0110 (-0.33)	-0.0114 $(-0.34)$	-0.0116 $(-0.34)$	-0.0116 $(-0.34)$	-0.0122 (-0.36)	-0.0126 (-0.37)	-0.0113 $(-0.34)$	-0.0112 (-0.33)	-0.0114 $(-0.34)$	-0.0131 $(-0.39)$
SALES GROWTHt-1	$\begin{array}{c} 0.0917^{***} \\ (8.08) \end{array}$	$\begin{array}{c} 0.0918^{***} \ (8.08) \end{array}$	$0.0920^{***}$ (8.10)	$\begin{array}{c} 0.0917^{***} \ (8.08) \end{array}$	$\begin{array}{c} 0.0917^{***} \\ (8.08) \end{array}$	$\begin{array}{c} 0.0918^{***} \ (8.08) \end{array}$	$\begin{array}{c} 0.0918^{***} \ (8.09) \end{array}$	$0.0918^{***} \\ (8.08)$	0.0918*** (8.08)	$0.0918^{***}$ $(8.08)$	$0.0916^{***}$ (8.07)
SIZEt-1	$-0.4301^{***}$ (-37.43)	$-0.4303^{***}$ (-37.47)	$-0.4307^{***}$ (-37.51)	-0.4305*** (-37.46)	$-0.4304^{***}$ (-37.49)	-0.4303*** (-37.48)	-0.4299*** (-37.43)	$-0.4305^{***}$ (-37.48)	$-0.4305^{***}$ (-37.49)	$-0.4303^{***}$ (-37.48)	-0.4299*** (-37.42)
BANK DEBTt-1	-0.4284*** (-20.33)	-0.4278*** (-20.34)	-0.4278*** (-20.33)	-0.4288*** (-20.19)	-0.4282*** (-20.34)	$-0.4278^{***}$ $(-20.34)$	-0.4273*** (-20.32)	-0.4280*** (-20.34)	-0.4281*** (-20.35)	-0.4278*** (-20.33)	-0.4284*** (-20.36)
Constant Crisis dummy Firm-fixed effects Nr. R <sup>2</sup>	yes yes yes 86,102 0.0913	yes yes 86,102 0.0914	yes yes 86,102 0.0913	yes yes 86,102 0.0912	yes yes 86,102 0.0913	yes yes 86,102 0.0913	yes yes yes 86,102 0.0913	yes yes 86,102 0.0912	yes yes yes 86,102 0.0913	yes yes yes 86,102 0.0912	yes yes yes 86,102 0.0913

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* weaker institutions). The dependent variable is INVESTMENT, defined as the flow of investment in tangible and intangible assets for each firm in a year over the total of those assets at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,09}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt at end-2009.  $LI_{i,09} \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010.

TABLE 8. Alternative approach: Firms' investment

		Fur	ding structu	re		Sovereign	exposure	Solve	ency	Funding	Collateral
	1	2	3	4	5	6	7	8	9	10	11
LI <sub>09</sub> :	Central bank	Interbank	Market	Dep_A	LTD	PT Sovereign	Sovereign	Tier 1 Capital	Total capital	Liq. Gap	Elegible A
LI <sub>09</sub> *CRISIS	$\begin{array}{c} 0.0244 \\ (0.42) \end{array}$	-0.0227*** (-2.72)	-0.0157** (-2.18)	$0.0140^{**}$ (2.07)	$\begin{array}{c} 0.0002 \ (0.63) \end{array}$	$0.1428^{***}$ (2.84)	$0.1448^{***}$ (2.78)	-0.0176 $(-0.97)$	-0.0255 $(-1.52)$	$\begin{array}{c} 0.0001 \\ (0.87) \end{array}$	$0.0539^{***}$ (2.71)
PROFITA BILITYt-1	$0.0321^{***}$ (3.87)	$\begin{array}{c} 0.0323^{***} \ (3.89) \end{array}$	$0.0322^{***}$ (3.88)	$0.0321^{***}$ (3.87)	$0.0321^{***}$ (3.87)	$0.0316^{***} (3.81)$	$\begin{array}{c} 0.0317^{***} \ (3.81) \end{array}$	$\begin{array}{c} 0.0322^{***} \ (3.88) \end{array}$	$\begin{array}{c} 0.0323^{***} \ (3.89) \end{array}$	$0.0320^{***}$ (3.86)	$\begin{array}{c} 0.0316^{***} \ (3.80) \end{array}$
SALES GROWTHt-1	$0.0492^{***} \\ (16.00)$	$\begin{array}{c} 0.0493^{***} \ (16.00) \end{array}$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.00)$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.01)$	$0.0493^{stst} \ (16.00)$	$0.0493^{***} \\ (16.00)$	$0.0493^{***}$ (16.00)	$\begin{array}{c} 0.0492^{***} \\ (15.98) \end{array}$
SIZEt-1	$-0.0625^{***}$ (-22.85)	-0.0625*** (-22.87)	-0.0626*** (-22.90)	-0.0624*** (-22.83)	-0.0625*** (-22.88)	$-0.0624^{***}$ (-22.84)	-0.0623*** (-22.79)	-0.0626*** (-22.88)	$-0.0626^{***}$ (-22.89)	-0.0626*** (-22.88)	$-0.0624^{***}$ (-22.81)
BANK DEBTt-1	-0.0444*** (-8.27)	-0.0443*** (-8.27)	-0.0443*** (-8.27)	$-0.0432^{***}$ (-8.02)	-0.0443*** (-8.26)	-0.0444*** (-8.27)	-0.0441*** (-8.23)	-0.0444*** (-8.28)	-0.0444*** (-8.29)	-0.0444*** (-8.27)	-0.0446*** (-8.31)
Constant	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Crisis dummy	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Firm-fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Nr.	86,304	86,304	86,304	86,304	86,304	86,304	86,304	86,304	86,304	86,304	86,304
R <sup>2</sup>	0.0967	0.0969	0.0969	0.0968	0.0968	0.0969	0.0969	0.0968	0.0968	0.0968	0.0969

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* weaker institutions). The dependent variable is EMPLOYMENT, defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,09}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt at end-2009.  $LI_{i,09} \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010.

TABLE 9. Alternative approach: Firms' employment

31

Summing up. In this analysis we explore the exposure of firms to adverse shocks through their lenders' positions defined at the end of 2009, before the onset of the sovereign debt crisis. In this Section, in particular, as the key variable is continuous and we analyze firm's outcomes in two periods (precrisis versus crisis), we can differentiate firms among those with high and low exposures to the negative market developments through their lenders' position. Thus, the interaction term between the Lenders' Indicator and the Crisis dummy can be compared to a treatment intensity variable. Consequently, even though the empirical approach adopted in this Section is not a "pure" differences-in-differences model, it retains the main features of that models.

Broadly, the main findings discussed in the previous sections continue to be valid under this empirical approach, such as the importance of lenders' funding positions. Additionally, the results highlight the relevance of collateral to overcome tensions in the financial markets, through the ECB monetary operations.

# 8. Robustness

In this study, we analyzed firms' outcomes exploring the heterogeneity on firm's lenders' exposure to the adverse developments in financial markets, in the context of the sovereign debt crisis. The variable of interest was *Lenders' Indicator*, measured by several alternative criteria. Following the specification presented in Section 7.1, we run some additional regressions, in order to assess how the results change due to some adopted hypotheses.

First, we replace the firm-fixed effects component  $(z_i)$ , in equation 3, with interaction terms between firms' business sector and the time dimension. These variables allow us to control, in some way, changes in firms' activity sector over time, which may influence firms' credit demand and decisions. In particular, we run the following specification:

$$y_{i,t} = c + \alpha X_{i,t-1} + \varphi Crisis + \beta LI_{i,09} \times Crisis + \delta S_i \times Crisis + \mu_{i,t} + \mu_{i,t}$$
(4)

where  $S_i \times Crisis$  corresponds to the interaction term between firm *i*'s business sector,  $S_i$ , and the time dummy *Crisis*. The remaining variables, both on righthand and left-hand sides in the equation, are defined as described in Section 7.1, namely:  $y_{i,t}$  corresponds to the average investment or average employment of firm *i* in the period *t* (2008/2009 and 2011/2012);  $X_{i,t-1}$  is a vector of firmspecific variables evaluated at t - 1;  $LI_{i,09}$  is the *Lenders' Indicator* based on all firm's lenders *i* at the end of 2009; while  $\mu_{i,t}$  corresponds to the error term. The results obtained under this set up are in line with those discussed above.

The next test comprises the inclusion of other lenders characteristics as explanatory variables in equation 3. Hence, in addition to the *Lenders*? *Indicator*, which remains the key variable in the analysis, we include measures related to leverage, profitability, and size of firms' lenders.<sup>26</sup>

In this robustness exercise, we run the following specification:

 $y_{i,t} = c + \alpha X_{i,t-1} + \delta z_i + \varphi Crisis + \beta LI_{i,09} \times Crisis + \varphi LC_{i,09} + \mu_{i,t}$ (5)

where  $LC_{i,09}$  is the vector of additional variables for lenders of firm *i*, assessed at the end of 2009 (lenders' control variables). The remaining variables included in the specification preserve the same definition as described in Section 7.1.

According to results obtained, the general conclusions continue broadly to hold under this conjecture. Nevertheless, there are some adjustments in the estimated coefficients related to *Lenders' Indicator*, but they are not sizable. The results obtained under this framework are shown in Tables A.7 and A.8, in the Appendix Section of this Chapter, for firms' investment and employment outcomes, respectively.

Therefore, even replacing firm-fixed effects by other firms' controls, or including some lenders' control variables, the empirical evidence continues to identify some impact on firms' decisions related to lenders' dependence on interbank and financial markets, as well as the availability of assets eligible as collateral in ECB operations. Looking at employment changes, in addition to these lenders' features, lenders' sovereign debt securities portfolios have also impact on firms' outcomes.

# 9. Final Remarks

Recent years have seen unprecedented events. First, the financial crisis related to the US sub-prime mortgage market, the collapse of Lehman Brothers, and the severe worldwide economic recession in 2009. Afterwards, the euro area faced several challenges due to the onset of the sovereign debt crisis in mid-2010. Those events dramatically affected the international financial markets and had consequences for financial systems. Indeed, following those events, financial intermediaries were forced to revise their funding and business strategies, taking into account a new and more demanding regulatory framework. Simultaneously, financial systems in some countries recorded additional constraints imposed by the international rescue programs. Against this background, the discussion on the transmission of the financial position of intermediary institutions to the real economy, in particular the corporate sector, became a topical issue in the economic and financial literature for academics and policy-makers.

33

<sup>26.</sup> The procedure adopted to compute lenders control variables at firm level followed the approach of *Lenders' Indicator*, *i.e.* a weighted average of firm's lenders' characteristics. The definitions of the control variables are presented in Table A.3 in the Appendix Section.

Portugal was one of the countries especially affected by the sovereign debt crisis, which led to the International Financial Assistance Programme in April 2011. The sovereign-bank link also had significant negative spillovers to the Portuguese banking institutions, which play a crucial role as financial intermediary for the Portuguese economy.

This study investigates the potential differences in firms' investment and employment decisions, taking into account all firm-bank relationships and the vulnerability of firms' lenders to the adverse financial market developments, in the context of the sovereign debt crisis. We compute the Lenders' Indicator, which is a weighted indicator for each firm based on lenders' characteristics and the share of each lender in the firm's total bank debt. Higher and lower Lenders' Indicators represent different vulnerabilities to the negative shock. In other words, we compare the path of investment and employment for the period 2007-2012, exploring firm's lenders' heterogeneity. The results obtained highlight the relevance lenders' funding structure. In particular, firms whose lenders depend more on market funding present a poorer investment and employment path. This suggests that there was some transmission of lenders' vulnerabilities to the corporate sector. This result is reinforced when we assess lenders' positions at the end of 2009, before the unexpected negative shock on sovereign debt markets. The results also show the relevance of eligible assets to pledge as collateral in monetary operations, measured at end-2009, in overcoming negative shocks. We found a positive impact of this indicator on average investment and employment decisions after 2010. However, based on other lenders' characteristics, namely the sovereign debt securities portfolios or even solvency position, the results are not so conclusive. While the results suggest that there was no impact on firms' investment decisions, there is some evidence of the impact of lenders' sovereign securities exposures on employment outcomes.

The different results observed in the investment and employment analysis may be related to the fact that investment reacts more quickly to the economic activity, uncertainty, or agents' confidence. Investment was already affected by the financial crisis and the economic recession in 2009. In turn, employment tends to present a higher lag of adjustment to adverse environments.

The empirical findings of this study suggest that the linkage of banking institutions and sovereigns could play an important role for corporate activity, and consequently for economic developments. The results also illustrate the need to improve the general economic analytical tools, taking into account the link between economic agents, financial system, and sovereigns. This is especially important for a small and open economy, mainly when it is highly dependent on international savings flows, such as the Portuguese case, which may amplify its exposure to international shocks.

# References

- Acemoglu, Daron and David Lyle (2004). "Women, war, and wages: The effect of female labor supply on the wage structure at midcentury." *Journal of political Economy*, 112(3), 497–551.
- Acharya, Viral V, Tim Eisert, Christian Eufinger, and Christian W Hirsch (2016). "Real effects of the sovereign debt crisis in Europe: Evidence from syndicated loans." Available at SSRN 2612855.
- Adelino, Manuel and Miguel A Ferreira (2016). "Bank ratings and lending supply: Evidence from sovereign downgrades." *Review of Financial Studies*, 29(7), 1709–1746.
- Adrian, Tobias, Paolo Colla, and Hyun Song Shin (2012). "Which Financial Frictions? Parsing the Evidence from the Financial Crisis of 2007-9." NBER Working Papers 18335, National Bureau of Economic Research, Inc.
- Almeida, Heitor, Igor Cunha, Miguel A Ferreira, and Felipe Restrepo (2017).
  "The Real Effects of Credit Ratings: The Sovereign Ceiling Channel." *Journal of Finance*, 72(1), 249–290.
- Almeida, Vanda and Ricardo Félix (2006). "Computing potential output and the output gap for the Portuguese economy." Economic Bulletin Autumn, Banco de Portugal.
- Amador, João and Carlos Coimbra (2007). "Characteristics of the Portuguese economic growth: what has been missing?" Working Paper w200708, Banco de Portugal.
- Angrist, Joshua D and Pischke, Jörn-Steffen (2008). "Parallel worlds: fixed effects, differences-in-differences, and panel data." In *Mostly harmless econometrics: An empiricist's companion*, chap. 5. Princeton university press.
- Bae, Kee-Hong, Jun-Koo Kang, and Chan-Woo Lim (2002). "The value of durable bank relationships: evidence from Korean banking shocks." *Journal of Financial Economics*, 64(2), 181–214.
- Banco de Portugal (2008). "Simplified reporting: Inclusion of the Simplified Corporate Information in the Statistics on Non-Financial Corporations from the Central Balance-Sheet Database." Supplements to the Statistical Bulletin, 1.
- Banco de Portugal (2011a). "'Central de Responsabilidades de Crédito'." Cadernos Nr.5, Banco de Portugal.
- Banco de Portugal (2011b). "Financial Stability Review." Tech. rep.
- Becker, Bo and Victoria Ivashina (2014). "Cyclicality of credit supply: Firm level evidence." Journal of Monetary Economics, 62, 76–93.
- Bentolila, Samuel, Marcel Jansen, Gabriel Jiménez, and Sonia Ruano (2015). "When Credit Dries Up: Job Losses in the Great Recession." Tech. rep.
- Benzoni, Luca, Pierre Collin-Dufresne, Robert S Goldstein, and Jean Helwege (2015). "Modeling credit contagion via the updating of fragile beliefs." *Review of Financial Studies*, 28(7), 1960–2008.

- Berger, Allen N. and Gregory F. Udell (2006). "A more complete conceptual framework for SME finance." Journal of Banking & Finance, 30(11), 2945–2966.
- Bernanke, Ben S (1983). "Nonmonetary Effects of the Financial Crisis in Propagation of the Great Depression." *American Economic Review*, 73(3), 257-76.
- Bernanke, Ben S, Mark Gertler, and Simon Gilchrist (1999). "The financial accelerator in a quantitative business cycle framework." In *Handbook of Macroeconomics*, vol. 1, edited by J. B. Taylor and M. Woodford, chap. 21, pp. 1341–1393. Elsevier.
- Bernanke, BS and M Gertler (1989). "Agency Costs, Net Worth, and Business Fluctuation." American Economic Review, 1989, 14–31.
- Bofondi, Marcello, Luisa Carpinelli, and Enrico Sette (2013). "Credit supply during a sovereign debt crisis." Temi di discussione (Economic working papers) 909, Bank of Italy, Economic Research and International Relations Area.
- Boot, Arnoud W. A. (2000). "Relationship banking: What do we know?" Journal of Financial Intermediation, 9(1), 7-25.
- Bottero, Margherita, Simone Lenzu, and Filippo Mezzanotti (2015). "Sovereign debt exposure and the bank lending channel: impact on credit supply and the real economy." Available at SSRN 2686226.
- Carvalho, Daniel, Miguel A. Ferreira, and Pedro Matos (2015). "Lending relationships and the effect of bank distress: Evidence from the 2007-2009 financial crisis." Journal of Financial and Quantitative Analysis, 50, 1165– 1197.
- Chodorow-Reich, Gabriel (2014). "The employment effects of credit market disruptions: Firm-level evidence from the 2008-09 financial crisis." *Quarterly Journal of Economics*, 129, 1–59.
- Cingano, Federico, Francesco Manaresi, and Enrico Sette (2013). "Does credit crunch investments down? New evidence on the real effects of the banklending channel." (91).
- De Marco, Filippo (2016). "Bank lending and the European sovereign debt crisis." Available at SSRN European 2673967.
- Diamond, Douglas W. (1984). "Financial intermediation and delegated monitoring." *Review of Economic Studies*, 51(3), 393-414.
- Fazzari, Steven, R Glenn Hubbard, and Bruce C Petersen (1987). "Financing constraints and corporate investment." Tech. rep., National Bureau of Economic Research.
- Holmstrom, Bengt and Jean Tirole (1997). "Financial intermediation, loanable funds, and the real sector." Quarterly Journal of Economics, 112(3), 663–91.
- Hoshi, Takeo, Anil Kashyap, and David Scharfstein (1990). "The role of banks in reducing the costs of financial distress in Japan." Journal of Financial Economics, 27(1), 67–88.

- Ivashina, Victoria and David Scharfstein (2010). "Bank lending during the financial crisis of 2008." Journal of Financial Economics, 97(3), 319–338.
- Iyer, Rajkamal, José-Luis Peydró, Samuel da Rocha-Lopes, and Antoinette Schoar (2014). "Interbank liquidity crunch and the firm credit crunch: Evidence from the 2007–2009 crisis." *Review of Financial Studies*, 27(1), 347–372.
- Kim, Moshe, Doron Kliger, and Bent Vale (2003). "Estimating switching costs: the case of banking." Journal of Financial Intermediation, 12(1), 25 56.
- King, Robert G and Ross Levine (1993). "Finance and growth: Schumpeter might be right." Quarterly Journal of Economics, 108(3), 717-737.
- Langfield, Sam and Marco Pagano (2015). "Bank bias in Europe: Effects on systemic risk and growth." Working Paper 1797, European Central Bank.
- Love, Inessa (2003). "Financial development and financing constraints: International evidence from the structural investment model." *Review of Financial studies*, 16(3), 765–791.
- Mark Gertler, Simon Gilchrist (1994). "Monetary policy, business cycles, and the behavior of small manufacturing firms." *Quarterly Journal of Economics*, 109(2), 309–340.
- Myers, S. (1984). "The capital structure puzzle." *Journal of Finance*, 39, 575–592.
- Ongena, Steven and David C. Smith (1998). chap. Quality and duration of bank relationships, pp. 224–235. Macmillan Press.
- Petersen, Mitchell A and Raghuram G Rajan (1994). "The benefits of lending relationships: Evidence from small business data." Journal of Finance, 49(1), 3–37.
- Popov, Alexander and Neeltje Van Horen (2015). "Exporting sovereign stress: Evidence from syndicated bank lending during the Euro Area sovereign debt crisis." *Review of Finance*, 19(5), 1825–1866.
- Rajan, Raghuram G (1992). "Insiders and outsiders: the choice between informed and arm's-length debt." Journal of Finance, 47(4), 1367–1400.

# $\mathbf{Appendix}$



FIGURE A.1: PT sovereign bonds (10 years)

Source: Thomson Reuters. This figure presents the Portuguese 10 year bond yields and the respective spread against comparable German bonds.



FIGURE A.2: Credit default swap spreads of Portuguese banks (5 years senior)

Sources: Bloomberg and Thomson Reuters.

This figure shows the CDS spreads of some major Portuguese banks, and the comparison with the CDS spread of iTraxx, that includes other banking institutions.



FIGURE A.3: Outstanding amounts of monetary policy operations of resident banks Source: Banco de Portugal.

This figure shows the credit obtained by resident banks in ECB monetary operations since 2008 and the first years of the sovereign debt crisis in the euro zone.



FIGURE A.4: Credit growth rate (year-on-year)

Source: Banco de Portugal.

This figure shows annual growth rate of bank loans granted to non-financial private sector from 2006 to 2014.

Variables	Definition
Dependent variables	
INVESTMENT GROWTH EMPLOYMENT GROWTH	Flow of investment in tangible and intangible assets for each firm in a year over the total of those assets at the end of the previous year Change in the average number of employees of each firm
	in a year over the average number in the previous year
Firm's characteristics	
PROFITABILITY	Net earnings before provisions and depreciation over total assets
SALES GROWTH	Bank debt over total assets Year-on-year change rate of sales
SIZE	Natural logarithm of real total assets
NUMBER RELATIONSHIPS	Number of banking relationships defined at the banking group level, based on the weight of each group in firm's total bank debt
Lenders' Indicator	
LI - LENDERS' INDICATOR	Interaction term between all firm-lenders' characteristics (based on indicators present below) and the respective weight in firm's total bank debt.

TABLE A.1. Variables definition - Firms

	Nr.	Mean	$\operatorname{Sd}$	p10	p25	p50	p75	p90
SIZE	211,741	13.78	1.45	12.11	12.82	13.63	14.58	15.63
PROFITABILITY	211,752	0.09	0.12	-0.01	0.04	0.09	0.14	0.22
SALES GROWTH	211,747	-0.03	0.26	-0.32	-0.14	-0.02	0.09	0.24
BANK CREDIT	211,752	0.66	0.31	0.30	0.47	0.66	0.81	0.93
ASSETS GROWTH	211,752	0.04	0.23	-0.18	-0.08	0.01	0.12	0.30
WORKING CAPITAL	211,752	0.16	0.35	-0.26	-0.02	0.18	0.39	0.57
ACCOUNTS PAYABLE	211,752	0.19	0.16	0.02	0.06	0.16	0.28	0.42
ACCOUNTS RECEIVABLE	211,752	0.28	0.22	0.00	0.08	0.25	0.43	0.59
TANGIBILITY	211,752	0.28	0.23	0.03	0.09	0.23	0.42	0.64
INVENTORIES	211,752	0.18	0.21	0.00	0.01	0.10	0.27	0.49
ASSET TURNOVER	211,752	1.33	1.01	0.41	0.70	1.09	1.65	2.47
CASHFLOW	211,752	0.06	0.26	-0.20	-0.03	0.07	0.18	0.33
CASH AND EQUIVALENTS	211,752	0.29	0.62	0.01	0.02	0.08	0.26	0.72
AGE	211,752	19.72	13.31	7.00	10.00	17.00	25.00	35.00
NUMBER RELATIONSHIPS	$211,\!752$	3.24	2.24	1.00	2.00	3.00	4.00	6.00

TABLE A.2. Sample summary statistics - Firm characteristics

Note: sd stands for standard deviation, while p10, p25, p50, p75, and p90 stand for the percentiles 10, 25, 50, 75, and 90, respectively, of the distribution of each variable, for observations included in the econometric analysis. Looking at the variables presented, SIZE is the natural logarithm of real total assets; PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; BANK CREDIT is defined as bank debt over total assets; SALES GROWTH is defined as Year-on-year change rate of sales; ASSETS GROWTH corresponds to the growth rate of total assets; WORKING CAPITAL is defined as current assets net of short term liabilities over total assets; ACCOUNTS RECEIVABLE is defined as accounts preceivable over total assets; ACCOUNTS RECEIVABLE is defined as accounts receivable over total assets; TANGIBILITY corresponds to tangible assets over total assets; INVENTORIES is defined as inventories over total assets; ASSET TURNOVER is defined as assets; CASHFLOW corresponds to the ratio of reachflow over total assets; CASH AND EQUIVALENTS is defined as cash and equivalents over total assets; AGE is defined in years. NUMBER RELATIONSHIPS reflects the number of banking relationships defined at the banking group level, based on the weight of each group in firm's total bank debt.

Variable	Definition
Lenders' Indicator	
Funding structure	
CENTRAL BANK INTERBANK MARKET FUNDING DEPOSITS_A LOAN-TO-DEPOSIT LIQUIDITY GAP	Central bank funding over total assets Interbank market over total assets Wholesale and interbank markets over total assets Customers' deposits over total assets Loans over customers' deposits Gap between assets and liabilities mismatch for 6-12 months
Debt Securities Portfolio	
PT SOVEREIGN SOVEREIGN	Portuguese sovereign debt securities over total assets Total sovereign debt securities over total assets
Collateral	
ELIGIBLE ASSETS	Assets eligible as collateral in monetary operations with central banks
Solvency	
TIER 1 RATIO CAPITAL RATIO	Tier 1 capital over risk weighted assets Total regular capital over risk weighted assets
Control variables	
LEVERAGE PROFITABILITY SIZE	Capital over total assets Returns on total assets Natural logarithm of total assets

TABLE A.3. Variables definition - Lenders

INVESTMENT	1.000												
EMPLOYMENT	0.145	1.000											
CENTRAL BANK	-0.053	-0.112	1.000										
INTER BANK	-0.015	-0.026	0.012	1.000									
MARKET	-0.026	-0.050	0.184	0.764	1.000								
LIQ. GAP	0.008	0.051	-0.264	-0.269	-0.338	1.000							
PT SOVEREIGN	-0.045	-0.100	0.608	-0.016	-0.024	-0.040	1.000						
SOVEREIGN	-0.049	-0.105	0.608	-0.016	0.059	-0.067	0.962	1.000					
EL. ASSETS	-0.037	-0.115	0.733	-0.001	0.073	-0.307	0.788	0.746	1.000				
R TIER1	-0.031	-0.055	0.299	0.032	0.187	0.242	0.404	0.470	0.379	1.000			
R CAPITAL	-0.028	-0.046	0.255	0.052	0.284	0.230	0.320	0.414	0.285	0.972	1.000		
$D\overline{E}POSITS A$	-0.059	-0.099	0.331	0.062	0.307	0.011	0.488	0.608	0.337	0.489	0.559	1.000	
LOAN-TO-DEP	-0.008	-0.006	0.007	0.263	0.197	-0.019	-0.007	-0.007	-0.003	0.078	0.079	-0.011	1.000

INVEST. EMPL. CENTRAL B. INTER B. MARKET LIQ. GAP PT SOV. SOV. EL ASSETS R\_TIER1 R\_CAPITAL DEP\_A LOAN-TO-DEP

TABLE A.4. Correlation Matrix: Investment, employment, and lenders' characteristics

Note: INVESTMENT is defined as the flow of investment in tangible and intangible assets for each firm in a year over the total of those assets at the end of the previous year. EMPLOYMENT is defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. The remaining variables characterize several dimensions on firm's lenders, namely: CENTRAL BANK is defined as the central bank funding over total assets; INTERBANK is defined as interbank funding over total assets; MARKET is defined as Wholesale and interbank markets over total assets; LIQ. GAP is defined as the gap between assets and liabilities mismatch for 6-12 months; SOV. PT is defined as the Portuguese sovereign debt securities over total assets; SOVEREIGN is defined as the total sovereign debt securities over total assets; EMPLOYMENT EL. ASSETS is defined as Assets eligible as collateral in monetary operations with central banks; R\_TIER1 corresponds to the Tier 1 capital over risk weighted assets; DEPOSITS\_A is defined as customers' deposits over total assets; LOAN-TO-DEP is defined as loans over customers' deposits.

			Funding s	Sovereig	Solvency				
	1	2	3	4	5	6	7	8	9
LI:	Central bank	Interbank	${ m M}{ m ar}{ m ke}{ m t}$	Dep_A	LTD	Sovereign	PT Sovereign	Tier 1 Capital	Total capita
$LI_{t-1}$	-0.0641 (-0.22)	-0.0204 $(-0.48)$	-0.0340 (-0.98)	$0.0170 \\ (0.49)$	-0.0004 (-0.50)	$\begin{array}{c} 0.2234 \ (0.54) \end{array}$	-0.1334 (-0.37)	-0.0406 (-0.21)	-0.0512 (-0.24)
$LI_{t-1}$ *2009	$egin{array}{c} 0.3183 \ (0.97) \end{array}$	$-0.0837^{*}$ $(-1.75)$	-0.0562 $(-1.32)$	$\begin{array}{c} 0.0455 \ (1.15) \end{array}$	-0.0011 $(-1.22)$	$\binom{0.7531}{(1.28)}$	$\begin{array}{c} 0.1470 \\ (0.35) \end{array}$	-0.0739 $(-0.36)$	-0.0503 $(-0.23)$
$LI_{t-1}$ *2010	-0.4261 (-1.04)	-0.0754 $(-1.35)$	-0.0651 $(-1.40)$	-0.0442 $(-1.03)$	-0.0019 $(-1.23)$	-0.4481 (-1.00)	-0.2260 (-0.56)	-0.1028 (-0.48)	-0.1187 (-0.52)
$LI_{t-1}$ *2011	$\begin{array}{c} 0.1132 \\ (0.37) \end{array}$	-0.1228** (-2.54)	-0.1137*** (-2.60)	-0.0642 $(-1.55)$	-0.0007 (-0.77)	-0.2676 (-0.67)	$\begin{array}{c} 0.0237 \\ (0.07) \end{array}$	-0.1046 $(-0.53)$	-0.1032 (-0.48)
$LI_{t-1}$ *2012	$egin{array}{c} 0.1889 \ (0.62) \end{array}$	$-0.1172^{**}$ (-2.55)	-0.1108*** (-2.64)	$\begin{array}{c} 0.0029 \\ (0.07) \end{array}$	-0.0010 (-1.20)	$\begin{array}{c} 0.0092 \\ (0.02) \end{array}$	$0.2918 \\ (0.81)$	$\begin{array}{c} 0.0002 \\ (0.00) \end{array}$	-0.0114 $(-0.05)$
PROFITA BILITYt-1	$\begin{array}{c} 0.0795^{***} \\ (3.22) \end{array}$	$0.0807^{***}$ (3.27)	$0.0802^{***}$ (3.25)	$0.0792^{***} \\ (3.21)$	$0.0801*** \\ (3.25)$	$0.0795^{***} (3.22)$	$0.0793^{***} \\ (3.21)$	$\begin{array}{c} 0.0794^{***} \\ (3.22) \end{array}$	$0.0794^{***}$ (3.22)
SALES GROWTHt-1	$0.1050*** \\ (14.54)$	$0.1046^{***}$ (14.48)	$0.1047^{***} (14.49)$	$0.1048^{***}$ (14.51)	$\begin{array}{c} 0.1048^{***} \\ (14.51) \end{array}$	$0.1049^{***}$ (14.51)	$\begin{array}{c} 0.1049^{***} \\ (14.52) \end{array}$	$\begin{array}{c} 0.1050^{***} \\ (14.54) \end{array}$	$0.1050^{***}$ (14.54)
SIZEt-1	-0.5164*** (-49.68)	-0.5152*** (-49.57)	$-0.5160*** \\ (-49.67)$	$-0.5164^{***}$ (-49.62)	-0.5162*** (-49.69)	-0.5163*** (-49.68)	-0.5164*** (-49.67)	-0.5165*** (-49.72)	-0.5165*** (-49.73)
BANK DEBTt-1	-0.3943*** (-22.77)	-0.3926*** (-22.69)	-0.3939*** (-22.76)	-0.3962*** (-22.78)	-0.3944*** (-22.80)	-0.3943*** (-22.72)	-0.3941*** (-22.70)	-0.3946*** (-22.79)	-0.3948*** (-22.81)
Constant Year dummies Firm-fixed effects Nr. R <sup>2</sup>	yes yes 211,752 0.0520	yes yes 211,752 0.0522	yes yes 211,752 0.0521	yes yes 211,752 0.0520	yes yes 211,752 0.0522	yes yes 211,752 0.0520	yes yes 211,752 0.0520	yes yes 211,752 0.0520	yes yes 211, 752 0.0520

#### TABLE A.5. Firms' investment decisions - yearly approach

LI:

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* more vulnerable institutions). The dependent variable is INVESTMENT, defined as the flow of investment in tangible and intangible assets for each firm in a year over the total of those assets at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,t-1}$  is the interaction term between the firm-lenders indicator under analysis in each column and the respective weight of each lender.  $LI \times$  year corresponds to the interaction between the previous variable and the year dummies (2008 to 2012).

		Fur	ding structu	re		Sovereig	n exposure	Solve	ency
	1	2	3	4	5	6	7	8	9
LI:	Central bank	Interbank	Market	Dep_A	LTD	Sovereign	PT Sovereign	Tier 1 Capital	Total capital
$LI_{t-1}$	-0.1384* (-1.91)	$0.0274^{***}$ (2.61)	$\begin{array}{c} 0.0136 \ (1.52) \end{array}$	-0.0209** (-2.33)	$0.0005^{**}$ (2.26)	-0.1535 (-1.28)	-0.1540 (-1.56)	-0.0724 (-1.34)	-0.0821 $(-1.40)$
$LI_{t-1}$ *2009	$\begin{array}{c} 0.1273 \ (1.52) \end{array}$	-0.0397*** (-3.13)	-0.0243** (-2.12)	$\begin{array}{c} 0.0163 \ (1.49) \end{array}$	-0.0005 $(-1.51)$	$\begin{array}{c} 0.0673 \ (0.40) \end{array}$	$0.0906 \\ (0.77)$	$egin{array}{c} 0.1176^{**} \ (2.03) \end{array}$	0.1255 ** (2.06)
$LI_{t-1}$ *2010	$0.1045 \\ (1.01)$	-0.0322** (-2.38)	$-0.0346^{***} (-2.99)$	$\begin{array}{c} 0.0147 \ (1.32) \end{array}$	-0.0001 (-0.22)	$\begin{array}{c} 0.1359 \ (1.07) \end{array}$	$egin{array}{c} 0.1057 \ (0.97) \end{array}$	$0.0596 \\ (1.03)$	$0.0638 \\ (1.05)$
$LI_{t-1}$ *2011	$0.0909 \\ (1.20)$	-0.0328*** (-2.66)	$-0.0315^{***}$ (-2.80)	$0.0243^{**}$ (2.23)	-0.0004* (-1.73)	$\begin{array}{c} 0.1882 \\ (1.64) \end{array}$	$0.1679^{*}$ (1.75)	$0.0508 \\ (0.92)$	$0.0536 \\ (0.91)$
$LI_{t-1}$ *2012	$0.1842^{**}$ (2.46)	$-0.0514^{***}$ (-4.23)	-0.0436*** (-3.82)	$\begin{array}{c} 0.0418^{***} \\ (3.80) \end{array}$	-0.0003 (-1.47)	$0.2913^{**}$ (2.49)	$egin{array}{c} 0.2972^{***}\ (3.01) \end{array}$	$0.1199^{**}$ (2.18)	$0.1216^{**}$ (2.06)
PROFITA BILITYt-1	$0.0809^{***}$ (12.94)	$0.0810^{***}$ (12.94)	0.0811*** (12.96)	0.0810*** (12.95)	$\begin{array}{c} 0.0808^{***} \\ (12.92) \end{array}$	0.0810*** (12.95)	0.0810*** (12.96)	0.0810*** (12.95)	0.0810*** (12.95)
SALES GROWTHt-1	$\begin{array}{c} 0.0367^{***} \ (17.35) \end{array}$	$\begin{array}{c} 0.0367^{***} \ (17.32) \end{array}$	$\begin{array}{c} 0.0367^{***} \\ (17.33) \end{array}$	$\begin{array}{c} 0.0367^{***} \ (17.35) \end{array}$	$0.0368*** \\ (17.37)$	$\begin{array}{c} 0.0367^{***} \ (17.34) \end{array}$	$\begin{array}{c} 0.0367^{***} \\ (17.33) \end{array}$	$\begin{array}{c} 0.0367^{***} \\ (17.35) \end{array}$	$\begin{array}{c} 0.0367^{***} \\ (17.35) \end{array}$
SIZEt-1	-0.0418*** (-19.51)	-0.0416*** (-19.39)	-0.0417*** (-19.46)	-0.0417*** (-19.41)	-0.0418*** (-19.49)	-0.0417*** (-19.43)	-0.0416*** (-19.41)	-0.0417*** (-19.48)	$-0.0418^{***}$ (-19.48)
BANK DEBTt-1	-0.0393*** (-8.96)	-0.0385*** (-8.80)	-0.0387*** (-8.83)	-0.0379*** (-8.63)	-0.0387*** (-8.84)	-0.0380*** (-8.67)	-0.0381*** (-8.69)	-0.0389*** (-8.89)	-0.0389*** (-8.89)
Constant Year dummies Firm-fixed effects N R <sup>2</sup>	yes yes 212,288 0.0397	yes yes 212,288 0.0398	yes yes 212,288 0.0397	yes yes 212,288 0.0397	yes yes 212,288 0.0397	yes yes 212, 288 0.0397	${ m yes} m yes$ yes 212,288 0.0397	${f yes}\ {f yes}\ {f yes}\ {f 212,288}\ 0.0398$	yes yes 212,288 0.0398

TABLE A.6. Firms' employment decisions - yearly approach

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* more vulnerable institutions). The dependent variable is EMPLOYMENT, defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,t-1}$  is the interaction term between the firm-lenders indicator under analysis in each column and the respective weight of each lender.  $LI \times$  year corresponds to the interaction between the previous variable and the year dummies (2008 to 2012).

		Fur	ding structu	re		Sovereign	exposure	Solve	ency	Funding	Collateral
	1	2	3	4	5	6	7	8	9	10	11
LI <sub>09</sub> :	Central bank	Interbank	Market	Dep_A	LTD	PT Sovereign	Sovereign	Tier 1 Capital	Total capital	Liq. Gap	Elegible A
$LI_{09}$ *CRISIS	$\begin{array}{c} 0.2364 \ (0.92) \end{array}$	-0.0662* (-1.87)	-0.0621** (-2.09)	-0.0096 (-0.36)	-0.0016 (-0.75)	$0.1549 \\ (0.74)$	$\begin{array}{c} 0.3000\ (1.36) \end{array}$	-0.0732 $(-0.64)$	-0.0993 (-0.96)	-0.0001 (-0.31)	$0.1485^{*}$ (1.68)
PROFITA BILITY t-1	-0.0123 $(-0.36)$	-0.0117 (-0.35)	-0.0120 (-0.36)	-0.0123 (-0.36)	-0.0122 (-0.36)	-0.0128 (-0.38)	-0.0130 (-0.39)	-0.0120 (-0.36)	-0.0119 $(-0.35)$	-0.0122 (-0.36)	-0.0135 $(-0.40)$
SALES GROWTHt-1	$\begin{array}{c} 0.0917^{***} \ (8.07) \end{array}$	$\begin{array}{c} 0.0918^{***} \ (8.08) \end{array}$	$\begin{array}{c} 0.0919^{***} \\ (8.09) \end{array}$	$\begin{array}{c} 0.0917^{***} \ (8.07) \end{array}$	$\begin{array}{c} 0.0917^{***} \\ (8.07) \end{array}$	$\begin{array}{c} 0.0917^{***} \ (8.08) \end{array}$	$\begin{array}{c} 0.0918^{***} \ (8.08) \end{array}$	$0.0917^{***}$ (8.07)	$\begin{array}{c} 0.0917^{***} \\ (8.07) \end{array}$	$\begin{array}{c} 0.0917^{***} \ (8.07) \end{array}$	$\begin{array}{c} 0.0916^{***} \ (8.06) \end{array}$
SIZEt-1	$-0.4299^{***} \\ (-37.41)$	$-0.4301^{***}$ (-37.45)	-0.4305*** (-37.48)	$-0.4302^{***}$ (-37.42)	$-0.4301^{***}$ $(-37.45)$	$-0.4301^{***}$ $(-37.44)$	-0.4298*** (-37.40)	$-0.4303^{***} (-37.43)$	$-0.4303^{***}$ (-37.44)	$-0.4301^{***}$ (-37.45)	-0.4298*** (-37.40)
BANK DEBTt-1	-0.4282*** (-20.32)	$-0.4277^{***}$ (-20.32)	-0.4275*** (-20.31)	-0.4285*** (-20.17)	-0.4281*** (-20.33)	-0.4277*** (-20.32)	-0.4271*** (-20.29)	-0.4282*** (-20.33)	-0.4285*** (-20.35)	-0.4278*** (-20.32)	-0.4281*** (-20.34)
Constant Crisis dummy	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes
Lenders controls Firm-fixed effects Nr.	yes yes 86102	$_{ m yes}^{ m yes}$ 86102	$_{ m yes}^{ m yes}$ 86102	$_{ m yes}^{ m yes}$ 86102	yes yes 86102	yes yes 86102	yes yes 86102	yes yes 86102	yes yes 86102	yes yes 86102	yes yes 86102
$\mathbb{R}^2$	0.0915	0.0915	0.0916	0.0914	0.0915	0.0915	0.0915	0.0915	0.0915	0.0914	0.0915

# TABLE A.7. Alternative approach: Firms' investment

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* weaker institutions). The dependent variable is EMPLOYMENT, defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,09}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt at end-2009.  $LI_{i,09} \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010.

		Fur	ding structu	re		Sovereign	exposure	Solve	ency	Funding	Collateral
	1	2	3	4	5	6	7	8	9	10	11
LI <sub>09</sub> :	Central bank	Interbank	Market	Dep_A	LTD	PT Sovereign	Sovereign	Tier 1 Capital	Total capital	Liq. Gap	Elegible A
$LI_{09}$ *CRISIS	$\begin{array}{c} 0.0174 \ (0.29) \end{array}$	-0.0254*** (-2.95)	-0.0202*** (-2.73)	$0.0150^{**}$ (2.22)	$\begin{array}{c} 0.0002 \ (0.53) \end{array}$	$\begin{array}{c} 0.1573^{***} \\ (3.04) \end{array}$	$0.1726^{***}$ (3.14)	$\begin{array}{c} 0.0408 \\ (1.42) \end{array}$	$\begin{array}{c} 0.0128 \ (0.50) \end{array}$	$0.0002^{**}$ (2.47)	$0.0699^{***}$ (3.29)
PROFITA BILITY t-1	$\begin{array}{c} 0.0322^{***} \ (3.88) \end{array}$	$\begin{array}{c} 0.0324^{***} \ (3.90) \end{array}$	$\begin{array}{c} 0.0323^{***} \ (3.89) \end{array}$	$0.0322^{***}$ (3.87)	$\begin{array}{c} 0.0322^{***} \ (3.88) \end{array}$	$\begin{array}{c} 0.0317^{***} \ (3.82) \end{array}$	$\begin{array}{c} 0.0317^{***} \ (3.82) \end{array}$	$0.0320^{***} (3.85)$	$0.0321^{***}$ (3.87)	$0.0320^{***} \\ (3.85)$	$\begin{array}{c} 0.0316^{***} \\ (3.81) \end{array}$
SALES GROWTHt-1	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.01)$	$0.0494^{***} (16.03)$	$0.0493^{***} \\ (16.02)$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.02)$	$0.0493^{***} \\ (16.02)$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.01)$	$0.0493^{***} \\ (16.01)$	$0.0492^{***}$ (15.99)
SIZEt-1	$-0.0625^{***}$ (-22.85)	-0.0625*** (-22.87)	-0.0626*** (-22.90)	$-0.0624^{***}$ (-22.82)	-0.0625*** (-22.87)	$-0.0624^{***}$ (-22.84)	-0.0623*** (-22.78)	$-0.0624^{***}$ (-22.83)	$-0.0625^{***}$ (-22.85)	$-0.0626^{***}$ (-22.90)	$-0.0623^{***}$ (-22.81)
BANK DEBTt-1	-0.0442*** (-8.24)	-0.0442*** (-8.23)	-0.0441*** (-8.23)	-0.0430*** (-7.97)	-0.0441*** (-8.23)	-0.0441*** (-8.23)	-0.0438*** (-8.17)	-0.0439*** (-8.18)	-0.0441*** (-8.21)	-0.0441*** (-8.23)	-0.0443*** (-8.27)
Constant Crisis dummy	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes	yes yes
Lenders controls Firm-fixed effects Nr. B <sup>2</sup>	yes yes 86304 0.0969	yes yes 86304 0.0971	yes yes 86304 0.0971	yes yes 86304 0.0970	yes yes 86304 0.0969	yes yes 86304 0.0971	yes yes 86304 0.0971	yes yes 86304 0.0969	yes yes 86304 0.0969	yes yes 86304 0.0970	yes yes 86304 0.0971

TABLE A.8. Alternative approach: Firms' employment

Note: \*\*\*, \*\*, and \* denote statistical significance levels at 1, 5, and 10 per cent, respectively. The t-statistics are in parentheses. The econometric models include firm-fixed effects and robust standard errors. All regressions include a constant term. Each column corresponds to an alternative criterion to classify firms' lenders as more vulnerable to sovereign debt crisis and its consequences (*i.e.* weaker institutions). The dependent variable is EMPLOYMENT, defined as change in the average number of employees for each firm in a year over the average number of employees at the end of the previous year. Looking at the independent variables, PROFITABILITY is defined as net earnings before provisions and depreciation over total assets; SALES GROWTH is defined as year-on-year change rate of sales; SIZE is based on the natural logarithm of real total assets; BANK DEBT defined as debt over total assets.  $LI_{i,00}$  is the interaction term between the firm-lenders indicator under analysis in each column and the weight of each lender in firm's debt at end-2009.  $LI_{i,09} \times Crisis$  corresponds to the interaction between the previous variable and the dummy variable that identifies the period after 2010.

# WORKING PAPERS

# 2015

 1|15
 Unpleasant debt dynamics: can fiscal consolidations raise debt ratios?

Gabriela Castro | Ricardo M. Félix | Paulo Júlio | José R. Maria

2|15 Macroeconomic forecasting starting from survey nowcasts

João Valle e Azevedo | Inês Gonçalves

3|15 Capital regulation in a macroeconomic model with three layers of default

Laurent Clerc | Alexis Derviz | Caterina Mendicino | Stephane Moyen | Kalin Nikolov | Livio Stracca | Javier Suarez | Alexandros P. Vardoulakis

4|15 Expectation-driven cycles: time-varying effects

Antonello D'Agostino | Caterina Mendicino

- 5|15 Seriously strengthening the tax-benefit link Pedro Portugal | Pedro S. Raposo
- 6|15 Unions and collective bargaining in the wake of the great recession

John T. Addison | Pedro Portugal | Hugo Vilares

- 7|15 Covariate-augmented unit root tests with mixed-frequency data Cláudia Duarte
- 8|15 Financial fragmentation shocks Gabriela Castro | José R. Maria | Paulo Júlio | Ricardo M. Félix
- 9|15 Central bank interventions, demand for collateral, and sovereign borrowing cost

Luís Fonseca | Matteo Crosignani | Miguel Faria-e-Castro 10|15 Income smoothing mechanisms after labor market transitions

Nuno Alves | Carlos Martins

- 11|15 Decomposing the wage losses of displaced workers: the role of the reallocation of workers into firms and job titles Anabela Carneiro | Pedro Raposo | Pedro Portugal
- 12|15 Sources of the union wage gap: results from high-dimensional fixed effects regression models

John T. Addison | Pedro Portugal | Hugo Vilares

13|15 Assessing european firms' exports and productivity distributions: the compnet trade module

Antoine Berthou | Emmanuel Dhyne | Matteo Bugamelli | Ana-Maria Cazacu | Calin-Vlad Demian | Peter Harasztosi | Tibor Lalinsky | Jaanika Meriküll | Filippo Oropallo | Ana Cristina Soares

14|15 A new regression-based tail index estimator: an application to exchange rates

João Nicolau | Paulo M. M. Rodrigues

- 15|15 The effect of bank shocks on firm-level and aggregate investment João Amador | Arne J. Nagengast
- 16|15 Networks of value added trade João Amador | Sónia Cabral
- 17|15 House prices: bubbles, exuberance or something else? Evidence from euro area countries

Rita Fradique Lourenço | Paulo M. M. Rodrigues

# 2016

- 1|16 A mixed frequency approach to forecast private consumption with ATM/POS data Cláudia Duarte | Paulo M. M. Rodrigues | António Rua
- 2|16 Monetary developments and expansionary fiscal consolidations: evidence from the EMU

António Afonso | Luís Martins

- 3|16 Output and unemployment, Portugal, 2008–2012 José R. Maria
  - <u>\_\_\_</u>
- 4|16 Productivity and organization in portuguese firms

Lorenzo Caliendo | Luca David Opromolla | Giordano Mion | Esteban Rossi-Hansberg

5|16 Residual-augmented IVX predictive regression

Matei Demetrescu | Paulo M. M. Rodrigues

- 6|16 Understanding the public sector pay gap Maria M. Campos | Evangelia Papapetrou | Domenico Depalo Javier J. Pérez | Roberto Ramos
- 7|16 Sorry, we're closed: loan conditions when bank branches close and firms transfer to another bank

Diana Bonfim | Gil Nogueira | Steven Ongena

8|16 The effect of quantitative easing on lending conditions

Laura Blattner | Luísa Farinha | Gil Nogueira

9|16 Market integration and the persistence of electricity prices

João Pedro Pereira | Vasco Pesquita | Paulo M. M. Rodrigues | António Rua

- 10|16 EAGLE-FLI | A macroeconomic model of banking and financial interdependence in the euro area
   N. Bokan | A. Gerali | S. Gomes | P. Jacquinot | M. Pisani
- 11|16 Temporary contracts' transitions: the role of training and institutions Sara Serra
- 12|16 A wavelet-based multivariate multiscale approach for forecasting António Rua
- 13|16 Forecasting banking crises with dynamic panel probit models

António Antunes | Diana Bonfim | Nuno Monteiro | Paulo M. M. Rodrigues

- 14|16 A tale of two sectors: why is misallocation higher in services than in manufacturing?Daniel A. Dias | Carlos Robalo Marques | Christine Richmond
- 15|16 The unsecured interbank money market: a description of the Portuguese case Sofia Saldanha
- 16|16 Leverage and risk weighted capital requirements Leonardo Gambacorta | Sudipto Karmakar
- 17|16 Surviving the perfect storm: the role of the lender of last resort Nuno Alves | Diana Bonfim | Carla Soares
- 18|16 Public debt expansions and the dynamics of the household borrowing constraint António Antunes | Valerio Ercolani



# 2017

 1|17 The diffusion of knowledge via managers' mobility
 Giordano Mion | Luca David Opromolla | Alessandro Sforza
 2|17 Upward nominal wage rigidity

> Paulo Guimarães | Fernando Martins | Pedro Portugal

**3|17** Zooming the ins and outs of the U.S. unemployment

Pedro Portugal | António Rua

- 4|17 Labor market imperfections and the firm's wage setting policy Sónia Félix | Pedro Portugal
- 5|17 International banking and cross-border effects of regulation: lessons from Portugal Diana Bonfim | Sónia Costa
- 6|17 Disentangling the channels from birthdate to educational attainment Luís Martins | Manuel Coutinho Pereira
- 7|17 Who's who in global value chains? A weighted network approach

João Amador | Sónia Cabral | Rossana Mastrandrea | Franco Ruzzenenti

8|17 Lending relationships and the real economy: evidence in the context of the euro area sovereign debt crisis

Luciana Barbosa

